# Terahertz Nonlinear Optics of Graphene: From Saturable Absorption to High-Harmonics Generation

Hassan A. Hafez, Sergey Kovalev, Klaas-Jan Tielrooij, Mischa Bonn, Michael Gensch, and Dmitry Turchinovich\*

Graphene has long been predicted to show exceptional nonlinear optical properties, especially in the technologically important terahertz (THz) frequency range. Recent experiments have shown that this atomically thin material indeed exhibits possibly the largest nonlinear coefficients of any material known to date, paving the way for practical graphene-based applications in ultrafast (opto-)electronics operating at THz rates. Here the advances in the booming field of nonlinear THz optics of graphene are reported, and the state-of-the-art understanding of the nature of the nonlinear interaction of graphene with the THz fields based on the thermodynamic model of electron transport in graphene is described. A comparison between different mechanisms of nonlinear interaction of graphene with light fields in THz, infrared, and visible frequency ranges is also provided. Finally, the perspectives for the expected technological applications of graphene based on its extraordinary THz nonlinear properties are summarized. This report covers the evolution of the field of THz nonlinear optics of graphene from the very pioneering to the state-of-the-art works. It also serves as a concise overview of the current understanding of THz nonlinear optics of graphene and as a compact reference for researchers entering the field, as well as for the technology developers.

### 1. Introduction

The advent of single-layer graphene<sup>[1-3]</sup> and ultrathin epitaxial graphite<sup>[4]</sup> in 2004 stimulated extensive research of the optical properties of graphene over a broad range of the electromagnetic spectrum.<sup>[5-12]</sup> These studies revealed universal absorption (or optical conductivity) at infrared (IR) and

Dr. H. A. Hafez, Prof. D. Turchinovich Fakultät für Physik Universität Bielefeld Universitätsstr. 25, 33615 Bielefeld, Germany E-mail: dmtu@physik.uni-bielefeld.de Dr. H. A. Hafez, Prof. M. Bonn Max-Planck-Institut für Polymerforschung Ackermannweg 10, 55128 Mainz, Germany

D The ORCID identification number(s) for the author(s) of this article can be found under https://doi.org/10.1002/adom.201900771.

© 2019 The Authors. Published by WILEY-VCH Verlag GmbH & Co. KGaA, Weinheim. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

#### DOI: 10.1002/adom.201900771

visible light frequencies, corresponding to interband transitions leading to electron-hole pair generation.<sup>[5,6,12]</sup> In the case of doped graphene, that is, graphene containing a certain concentration of free electrons or holes, Drude-like conductivity was observed at far-infrared and terahertz (THz) frequencies, corresponding to intraband free-carrier absorption.<sup>[8,9,12–15]</sup> Further, theoretical studies predicted strong nonlinear interaction of graphene with intense light fields, in particular at the technologically important THz frequencies (typically, in the range 0.1–10 THz),<sup>[16–25]</sup> as originating from both interband and intraband electron dynamics. These predictions were inspired by the unique band structure of graphene: absence of a bandgap and linear energy-momentum dispersion for its electrons.<sup>[2,3,26–28]</sup> A plethora of strong nonlinear effects in graphene in the IR and optical frequency ranges, originating from interband electron dynamics, was successfully demonstrated, including

saturable absorption and nonlinear refraction,<sup>[29–39]</sup> higherharmonic generation,<sup>[40–47]</sup> and wave-mixing processes<sup>[48–50]</sup> (see also reviews<sup>[51–53]</sup>). At THz frequencies, however, until recently only saturable absorption effects in doped graphene,<sup>[54–61]</sup> and induced multiphoton-like absorption in multilayer near-intrinsic graphene were successfully demonstrated.<sup>[62]</sup> At the same time, the observation of the long sought-after effect of THz higher-order harmonics generation,

Helmholtz-Zentrum Dresden-Rossendorf Bautzner Landstraße 400, 01328 Dresden, Germany Dr. K.-J. Tielrooij Catalan Institute of Nanoscience and Nanotechnology (ICN2) BIST & CSIC Campus UAB, Bellaterra 08193, Barcelona, Spain Prof. M. Gensch Institut für Optische Sensorsysteme DLR Rutherfordstraße 2, 12489 Berlin, Germany Prof. M. Gensch Institut für Optik und Atomare Physik Technische Universität Berlin Strasse des 17. Juni 135, 10623 Berlin, Germany

Dr. S. Kovalev



that is, the multiplication of the frequency of incident THz photons in graphene, remained elusive. This situation has changed recently when very efficient generation of THz harmonics up to the seventh order in single-layer-doped graphene was demonstrated using multicycle quasi-monochromatic THz driving fields, revealing gigantic nonlinear optical coefficients of graphene in the THz range.<sup>[63]</sup> The physical mechanism of the demonstrated highly efficient THz high-harmonics generation (THz HHG) and THz saturable absorption in doped graphene was attributed to the collective thermodynamic response of the background Dirac electrons of the doped graphene to the THz driving field, which is currently believed to be the dominating effect in interaction of free electrons in graphene with THz fields. The underlying physical picture of this interaction, and its effect on the THz conductivity and absorbance in graphene will be provided within this report. Finally, we emphasize the importance of exploring the THz nonlinear properties of graphene for emerging technologies. Graphene-based transistors,[64-67] photodetectors,<sup>[68,69]</sup> saturable absorbers for laser modelocking,<sup>[30,70,71]</sup> and other high-speed (opto-)electronic devices, all usually operate in the nonlinear regime, and are entering the THz frequency/rate operation mode.

#### 1.1. How Is This Report Organized?

Our report is organized in five sections. After the introduction in section 1, section 2 provides a brief overview of the important fundamentals of graphene, such as its band structure and key electronic properties. Section 3 deals with the linear interaction of graphene with light fields in general, accounting for the interband and intraband electron dynamics. Section 4 presents (i) the experimental observations of nonlinear THz optics of graphene and other graphene-based systems, and a brief overview of the experimental techniques and facilities for the generation of intense THz fields, (ii) the thermodynamic model that describes the nonlinear interaction of graphene with intense THz fields, and (iii) the experimental generation of THz high harmonics, which directly follows from the thermodynamic picture of THz electron transport in graphene. Where relevant, the results on graphene under optical excitation will also be briefly discussed. Finally, section 5 provides conclusions and future perspectives on possible applications of graphene as a highly nonlinear THz material.

# 2. Band Structure of Graphene

Graphene is a monolayer of carbon atoms bound covalently in a hexagonal honeycomb lattice, as shown in **Figure 1a**. Each carbon atom in the graphene plane is bound to three nearest neighbor atoms via strong equivalent  $\sigma$ -bonds established by sp<sup>2</sup> hybridization of three of the valence electron-orbitals of each atom, namely, 2s, 2p<sub>x</sub>, and 2p<sub>y</sub>,<sup>[26–28,72]</sup> as shown in Figure 1b. These in-plane  $\sigma$ -bonds are responsible for the mechanical strength of graphene. The fourth electron-orbital 2p<sub>z</sub> is oriented perpendicular to the graphene plane. The overlap between the out-of-plane 2p<sub>z</sub> orbitals of the carbon atoms results in weaker





Hassan A. Hafez obtained his M.Sc. degree from Helwan University, Egypt, in 2008, and his Ph.D. degree from the University of Quebec (INRS-EMT), Canada, in 2016. Then, he became a lecturer and faculty staff member at Helwan University, followed by a postdoctoral stay at MPI for Polymer Research in Germany, and a position as research

associate at the University of Duisburg-Essen. In 2019, he was appointed *akademischer Rat* at Bielefeld University where he is currently leading the development of high-intensity terahertz facility. His research interests include nonlinear terahertz optics and ultrafast dynamics in condensed matter.



Michael Gensch obtained his Ph.D. degree from the TU Berlin in 2004. After postdoctoral stays at the Leibniz-Institute of Analytical Sciences and Deutsches Elektronen-Synchrotron and a position as staff scientist at the Helmholtz-Zentrum Berlin, he joined the Helmholtz-Zentrum Dresden-Rossendorf as project leader for the TELBE

Terahertz facility in 2010. In 2015, he became group leader for "High-Field Terahertz-Driven Phenomena". In 2019, he became professor of physics at the TU Berlin and head of the "Terahertz and Laser Spectroscopy" department at the DLR Institute of Optical Sensor Systems. His research interests include terahertz spectroscopy, ultrafast science, instrumentation development, and 2D materials.



#### Dmitry Turchinovich

graduated from St. Petersburg Electrotechnical University and loffe Institute in 1999 and obtained his Ph.D. from the University of Freiburg in 2004. After a postdoctoral stay at Utrecht University, in 2006 he joined the faculty at Technical University of Denmark. In 2012–2017 Dmitry Turchinovich led the Ultrafast Dynamics &

Terahertz Spectroscopy group at MPI for Polymer Research in Mainz. In 2017 he was appointed professor of physics at the University of Duisburg-Essen, and in 2018 he became full professor of physics at Bielefeld University. His research interests are ultrafast and terahertz dynamics in condensed matter, and general ultrafast science.





**Figure 1.** Graphene lattice and band structure: a) Hexagonal honeycomb lattice of graphene, b) binding between the carbon atoms in the graphene lattice, c) the triangular sublattices *A* and *B*, d) the first hexagonal Brillouin zone exhibiting two inequivalent groups of Dirac points, and e) the energy-momentum spectrum of the graphene band structure described by Equation (1). Note the rotation of *k*-axes with respect to (d). (Reproduced, except for (b), with permission.<sup>[26]</sup> Copyright 2009, American Physical Society).

 $\pi$ -bonds. These delocalized  $\pi$ -electrons are responsible for the electrical conduction in graphene.

The hexagonal lattice of graphene is considered as a bipartite lattice composed of two triangular sublattices, marked A and B in Figure 1c. The sublattices are the Bravais lattices of graphene with lattice vectors  $a_1 = 0.5a(\sqrt{3}, 1)$  and  $a_2 = 0.5a(\sqrt{3}, -1)$  and a lattice constant  $a = |a_1| = |a_2| = 0.246$  nm.<sup>[27]</sup> We note here that the lattice constant *a* is related to the carbon–carbon distance  $a_{cc} = 0.142$  nm by  $a = \sqrt{3} a_{cc}$ .

In the reciprocal space, the first Brillouin zone is consequently a hexagon featuring six corner points of two inequivalent groups of *K*-points called the *Dirac points* indicative of the two sublattices A and B, as shown in Figure 1d. The corresponding band structure is shown in Figure 1e, in which the valence band touches the conduction band at these six Dirac points, providing a gapless band structure of graphene. This band structure of graphene was first calculated by Wallace in 1947 using the tight-binding approximation,<sup>[73]</sup> providing cosine-like energy bands in energy–momentum space<sup>[27]</sup>

$$\varepsilon(k_x, k_y) = \pm \gamma_0 \left[ 1 + 4\cos\left(\frac{\sqrt{3}k_x a}{2}\right) \cos\left(\frac{k_y a}{2}\right) + 4\cos^2\left(\frac{k_y a}{2}\right) \right]^{1/2}$$
(1)

where  $\gamma_0 \approx 2.8$  eV is the hopping energy (transfer integral) between the nearest-neighbor atoms. Near the Dirac points where  $|k|a \ll 1$ , the Taylor series expansion of the energy spectrum of Equation (1) reduces to the linear dispersion relation<sup>[27]</sup>

$$\varepsilon_{\pm}(k) = \pm \frac{\sqrt{3}}{2} \gamma_0 a k = \pm \nu_F \hbar k \tag{2}$$

where  $v_F = \sqrt{3} \gamma_0 a/2\hbar \approx 10^6 \text{ m s}^{-1}$  is the energy- and momentumindependent band velocity (Fermi velocity)<sup>[73]</sup> and the ± sign is indicative of the positive kinetic energy of electrons in the conduction band (the  $\pi^*$  electrons) and the negative (binding) energy of the bound electrons (the  $\pi$  electrons) in the valence band. This unique linear dispersion  $\varepsilon = v_F \hbar k$  with constant band velocity in the vicinity of the Dirac points in graphene is characteristic of ultrarelativistic massless quasiparticles: Dirac fermions with zero rest mass. The linear band structure approximation of Equation (2) is valid for the energies up to about  $\pm 1.5$  eV with respect to the Dirac point, thus covering a very broad range of energies (frequencies) of the electromagnetic spectrum from THz to UV.

The density of states in graphene  $D(\varepsilon)$  within the linear part of the band structure is a linear function of energy, with D = 0at the Dirac point  $\varepsilon = 0$ 

$$D(\varepsilon) = \frac{2}{\pi \hbar^2 v_{\rm F}^2} |\varepsilon| \tag{3}$$

The Fermi energy  $E_{\rm F}$  depends on the free carrier density  $N_{\rm c}$  in graphene as

$$E_{\rm F} = \hbar v_{\rm F} \sqrt{\pi N_{\rm c}} \tag{4}$$

Very importantly for the electronic transport in graphene, it can be shown that for the "massless" electrons with linear energy-momentum dispersion such as Equation (2) it is *fundamentally* not possible to electrostatically bind them to a particle of opposite charge. This is in stark contrast to the "massive" charge carriers with quasi-parabolic energy-momentum dispersion, such as electrons and holes in regular metals or semiconductors, which can be trapped by charged impurities, or form excitons. This absence of electrostatic (Coulomb) binding for its electrons also makes graphene stand out among other 2D materials with "massive" quasi-parabolic energy momentum dispersion for the charge carriers, such as transition metal dichalcogenides, which feature extremely strong electronic Coulomb binding due to the very weak dielectric (lattice) screening, leading in some cases to exciton binding energies as high as  $\approx 500 \text{ meV.}^{[74]}$ 

ADVANCED SCIENCE NEWS \_\_\_\_\_

In graphene, due to the fundamental absence of Coulomb binding, electron transport is not affected by the trapping onto impurities and other unscreened Coulomb potentials, and is only limited by the momentum scattering. This unique property of graphene electrons, combined with the large optical phonon energy of graphene of the order of ~200 meV, and the nonpolar nature of its lattice, leads to a very large carrier mobility that can exceed 100 000 cm<sup>2</sup> V<sup>-1</sup> s<sup>-1</sup> observable even at room temperature.<sup>[66,72,75]</sup>

Further, graphene exhibits a strong ambipolar field effect when subjected to a gating voltage, allowing for a precise control of the type and density of the free charge carriers (electrons or holes) in the respective bands.<sup>[1]</sup> Combined with naturally high electron mobility, this made graphene a promising candidate for the development of ultrafast transistors operating at the rates of 100s of GHz.<sup>[64,65]</sup>

# 3. Light-Matter Interaction in Graphene in the Linear Regime

Depending on the photon energy  $\hbar\omega$  and the Fermi energy of graphene  $E_{\rm F}$ , the excitation of graphene with light will lead to either interband or intraband transitions. The pioneering observation and distinction between both types of transitions in graphene was made by Winnerl et al.<sup>[9]</sup> Optical excitation of graphene with  $\hbar \omega > 2E_{\rm F}$  results in an interband transition, similar to photoexcitation of semiconductors, and leads to the creation of an electron-hole pair, as shown schematically in Figure 2a. The interband transitions in graphene exhibit a universal, that is, frequency-independent, absorption  $A = \pi \alpha = 2.3\%$  of the incident light,<sup>[5]</sup> where  $\alpha = (1/4\pi\epsilon_0) e^2/\hbar c = 1/137$  is the finestructure constant, corresponding to the universal optical conductivity  $\sigma_0 = e^2/4\hbar \equiv 1$  G. Such a universal light absorption is a direct consequence of the unique electronic properties of graphene: the linear energy-momentum dispersion and the gapless band structure. We note that for photon energies below  $\approx 0.5$  eV and above  $\approx 2$  eV, a deviation from this universal absorption occurs.<sup>[6,12,76]</sup> The frequency dependence of the corresponding interband conductivity was shown to follow the relation[6,8-10,12]

$$\sigma_{\text{inter}}(\omega) = \frac{e^2}{8\hbar} \left[ \tanh\left(\frac{\hbar\omega + 2E_{\text{F}}}{4k_{\text{B}}T_{\text{e}}}\right) + \tanh\left(\frac{\hbar\omega - 2E_{\text{F}}}{4k_{\text{B}}T_{\text{e}}}\right) \right]$$
(5)

where  $k_{\rm B}$  is the Boltzmann constant and  $T_{\rm e}$  is the electron temperature. We note that for large photon energies, that is,  $\hbar \omega \gg 2E_{\rm F}$  and  $\hbar \omega \gg k_{\rm B}T_{\rm e}$ , Equation (5) reduces to the universal conductivity  $\sigma_0 = e^2/4\hbar$ . We further note that the optical absorption (and reflection) of graphene depends strongly on its dielectric environment.<sup>[77,78]</sup>

For photons of energy  $\hbar \omega < 2E_{\rm F}$ , as is typically the case at THz frequencies, the interband transitions are Pauliblocked,<sup>[9,12,79–81]</sup> whereas the intraband transitions are possible, as depicted schematically in Figure 2b. Such transitions correspond to the interaction of the incident light with the electrons around the Fermi surface of graphene, that is, to the free carrier absorption. The frequency-dependent optical conductivity corresponding to such intraband transitions, in case of a thermally equilibrated electron population, is described by the following solution of the Boltzmann transport equation<sup>[28,59]</sup>

$$\tilde{\sigma}_{\text{intra}}(\omega) = -\frac{e^2 v_{\text{F}}^2}{2} \int_{0}^{\infty} D(\varepsilon) \frac{\tau(\varepsilon)}{1 - i\omega\tau(\varepsilon)} \frac{\partial f_{\text{FD}}(\varepsilon, \mu, T_{\text{e}})}{\partial \varepsilon} d\varepsilon$$
(6)

where  $\tau(\varepsilon)$  is the energy-dependent electron momentum scattering time, and  $f_{\rm FD}(\varepsilon, \mu, T_{\rm e})$  is Fermi–Dirac distribution function depending on the electron energy  $\varepsilon$ , chemical potential  $\mu$ , and electron temperature  $T_{\rm e}$ .

For the case of  $T_{\rm e} \ll E_{\rm F}/k_{\rm B}$ , which corresponds to a sharp Fermi edge, only the electron scattering time at the Fermi energy,  $\tau = \tau(E_{\rm F})$ , matters, and Equation (6) reduces to the following Drude-type expression<sup>[8,9,82]</sup>

$$\tilde{\sigma}_{intra}\left(\omega\right) = \frac{2e^{2}\tau}{\pi\hbar^{2}\left(1 - i\omega\tau\right)}k_{\rm B}T_{\rm e}\ln\left[2\cosh(E_{\rm F}/2k_{\rm B}T_{\rm e})\right]$$
(7)

The THz excitation of graphene is dominated by the intraband conductivity (Equations (6) and (7)), especially at room temperature. Indeed, the photon energy in the few meV range (1 THz = 4.1 meV) is considerably smaller than the roomtemperature thermal energy  $k_{\rm B}T_{\rm e} = 26$  meV. As a result, in intrinsic, undoped graphene with  $E_{\rm F} \approx 0$ , at room temperature, the thermal population of the conduction and valence bands becomes almost equal: ( $f_{FD, \text{ conduction}} - f_{FD, \text{valence}}$ ) |<sub>1THz, 300K</sub> = 0.04 for the interband transition at the frequency of 1 THz = 4.1 meV. This leads to effective Pauli-blocking of the interband THz transitions in graphene at room temperature, making them only possible under cryogenic conditions (see, e.g., ref. [62]). However, in practical THz experiments on large area, ≈mm sized, single-layer graphene samples, even at lower temperatures the THz interband transitions are not easy to reveal because of the presence of so called "charge puddles," local areas of n- or p-doping with  $|E_{\rm F}| > 0$ , again Pauli-blocking the interband THz transitions.<sup>[83]</sup>

For the typical doping levels for substrate-supported graphene, resulting in a Fermi energy on the order of 100 meV (see, e.g., refs. [59,63,72,84–90]), the interband THz transitions are Pauli-blocked even at T = 0, leaving only the intraband transitions described by the free carrier conductivity of Equation (6) possible.

We note that most single-layer graphene samples are in fact unintentionally doped due to impurities induced in the preparation process as well as the effects of the environment in which graphene is used.<sup>[72,85-90]</sup> For example, in the course of preparation of chemical vapor deposition (CVD) grown graphene, which is one of the most popular graphene forms broadly available nowadays, hole-doping is induced during the transfer of poly(methyl methacrylate) (PMMA)-covered graphene from copper foils to dielectric substrates by residues of PMMA.<sup>[90]</sup> The dielectric substrate itself, and water and oxygen molecules trapped between the graphene and the substrate further contribute to the hole-doping of CVD graphene.<sup>[90,91]</sup> On the other hand, in epitaxially grown graphene on SiC electrondoping is induced by the SiC substrate (substrate-induced cha rging).<sup>[9,27,86,92,93]</sup> For multilayer epitaxial graphene, the layers closer to the middle of the sample can be intrinsic with  $E_{\rm F} \approx 0$ , while outer layers are usually doped by the environment and the substrate.<sup>[9,62,93]</sup>

www.advancedsciencenews.com

ADVANCED OPTICAL MATERIALS www.advopticalmat.de



**Figure 2.** Interaction of graphene with a light field: Schematic of the band structure with a) interband transition due to absorption of a photon of energy larger than twice the Fermi energy,  $\hbar \omega > 2E_F$ , leading to electron-hole pair generation, followed by the energy and momentum exchange with the Fermi sea electrons via electron-electron scattering, and b) intraband dynamics when  $\hbar \omega < 2E_F$ , which exhibits Drude-like conductance. c) Demonstration of the transmission of a THz field through a graphene sample with absorbed portion of the field  $\Delta E_{THz}$  related to the graphene intraband conductivity (Reproduced under the terms of the Creative Commons CC-BY 4.0 License.<sup>[59]</sup> Copyright 2015, Springer Nature). d,e) Actual measurement of THz fields and the corresponding spectra, respectively, transmitted through a graphene sample on SiO<sub>2</sub> substrate and through a bare SiO<sub>2</sub> substrate as a reference, and f) the corresponding complex THz intraband conductivity normalized to the universal conductivity of Equation (7) using  $E_F = 170$  meV and  $\tau = 47$  fs as fitting parameters (Reproduced with permission.<sup>[63]</sup> Copyright 2018, Springer Nature).

Given the symmetry of Dirac band structure of graphene (see Figure 1), there is no fundamental difference between the electron and hole doping: Both charge species can be considered equivalent and conduct electric currents in the same fashion. In the literature, for simplicity, the Fermi level is often indicated within the conduction band even if graphene is in fact hole-doped (such as typical CVD-grown graphene). Correspondingly, irrespective of the actual type of majority carriers in graphene, they are often referred to as electrons. In this paper, we follow this convention, unless specifically indicated.

Figure 2c–f shows the THz field transmission through a monolayer CVD-grown p-doped graphene sample, and the corresponding THz power transmission and conductivity

spectra of graphene. With THz time-domain spectroscopy (THz-TDS),<sup>[94,95]</sup> which is based on direct time-domain sampling of the propagating THz electric field, one can extract the graphene conductivity from the experimentally measured THz field transmission of the graphene film through the following relation in the frequency domain (known as Tinkham equation)<sup>[96]</sup>

$$\tilde{T}(\omega) = \frac{\tilde{E}_{s}(\omega)}{\tilde{E}_{ref}(\omega)} = \frac{n_{s}+1}{n_{s}+1+Z_{0}\tilde{\sigma}(\omega)}$$
(8)

where  $n_{\rm s}$  is the refractive index of the substrate at the THz frequencies,  $Z_0 = 377 \ \Omega$  is the free-space impedance, and  $\tilde{E}_{\rm s}(\omega)$  and  $\tilde{E}_{\rm ref}(\omega)$  are the spectral complex Fourier transforms of the

## www.advancedsciencenews.com

Table 1. Sim	plified overview of the	e fundamental timescale	of optical fields and	electron dynamics in nor	intrinsic ( $E_F > 0.1 \text{ eV}$ ) graphene.
--------------	-------------------------	-------------------------	-----------------------	--------------------------	--

Spectral range and corresponding typical light-wave frequency and wavelength	Light field oscillation period	Electron transport lifetime (average electron momentum scattering time)	Electron-electron scattering (thermalization time)	Electron–SCOP <sup>a)</sup> scattering	Electron-phonon scattering
NIR: 200 THz, 1.5 μm	5 fs	10–100 fs	10–100 fs	10–100 fs	≈1 ps
THz: 1 THz, 300 μm	1 ps				

<sup>a)</sup>SCOP = strongly coupled optical phonons.

measured time-domain fields  $E_s(t)$  and  $E_{ref}(t)$ , respectively, as shown in Figure 2d. Here,  $E_s(t)$  is the THz field transmitted through graphene on a substrate, and  $E_{ref}(t)$  is that transmitted through a bare substrate of identical thickness used as a reference. Figure 2f shows the complex conductivity, real and imaginary, obtained from the experimental THz fields shown in Figure 2d at room temperature, and fitted by the Drude-like conductivity of Equation (7). The only adjustable parameters in this fit are the Fermi energy  $E_{\rm F}$  and the momentum scattering time  $\tau$ . Thus, THz-TDS allows one to unambiguously determine the key parameters of electron transport in graphene in the linear regime-the number density of conducting charge  $N_{\rm c}$ , linked to  $E_{\rm F}$  by Equation (4), and the average momentum charge carrier scattering time  $\tau$ . Note that the *sheet* conductivity (i.e., conductivity times thickness) in units of  $S = \Omega^{-1}$  is used in Equation (8).

#### 4. Terahertz Nonlinear Interactions in Graphene

# 4.1. General Considerations on THz and Optical/IR Nonlinearities in Graphene

Since the advent of graphene, several theories were put forward that predict a highly nonlinear response of this material, in particular in the THz regime. This field of research was initiated

by Mikhailov,[16,17,19] who initially used semiclassical calculations and ballistic transport. Later theories were extended to include quantum mechanical considerations,<sup>[18,21,22]</sup> and more recent theories of graphene nonlinearities furthermore take into account realistic conditions regarding momentum scattering time and ambient temperature. Interestingly, the effect of the elevated electron temperature on graphene nonlinearities was not considered of significant importance in these earlier works. In 2015, Mics et al.<sup>[59]</sup> showed that the strong THz nonlinearity in graphene is the result of THz-induced carrier heating, which was followed by several other studies, culminating in the observation of extremely efficient THz high-harmonics generation in graphene in 2018 by Hafez et al.<sup>[63]</sup> Regarding the visible and near-infrared (NIR) regime, the role of the increased carrier temperature on graphene nonlinearity was first addressed in 2018 by Soavi et al.[46] and Baudisch et al.<sup>[97]</sup> These studies show that properly taking into account the electron temperature leads to a significant adjustment in the theoretically predicted nonlinearities, yielding excellent agreement between theory and experiment.

The nonlinearity in graphene is naturally related to the transport of its electrons in the applied electric field of an incident light wave. In order to understand the fundamental difference between graphene nonlinearities in the THz regime and in the optical or NIR regime, we first summarize the relevant timescales of incident light wave oscillations and of electron dynamics in graphene (see **Table 1** and **Figure 3**).



**Figure 3.** Illustration of nonlinear graphene–light interaction at NIR and THz frequencies: a) Illustration of the nonlinear graphene electron current response (red) from incident near-infrared (NIR) light at 200 THz (blue), after Baudisch et al.<sup>[97]</sup> The dominant dynamics are based on the coherent photon–electron interactions. Since the induced electron current does not exactly follow the incident electric field, redshifted harmonics are generated. b) Illustration of the nonlinear graphene heat response (red) from incident THz light at 1 THz (blue), after Hafez et al.<sup>[63]</sup> The dominant dynamics are based on the heating–cooling dynamics of the graphene–electron system, which modulate the graphene intraband conductivity and consequently the induced electron current, leading to re-emission at THz higher harmonics. The lengths of the black lines under the figure panels indicate the fundamental timescales of the relevant processes, scaling with the corresponding figure above them.

1900771 (6 of 25)

We remind that since Coulomb trapping for massless Dirac fermions is fundamentally forbidden, electronic transport in graphene is dominated by momentum scattering events. Graphene electrons undergo momentum scattering by longrange and short-range interactions with charged impurities, disorder, and phonons.<sup>[28]</sup> In substrate-supported graphene, the former typically dominates, and the momentum scattering time is typically on the order of 10s fs, corresponding to mobilities in the range of 1000–10 000 cm<sup>2</sup> V<sup>-1</sup> s<sup>-1</sup>. If the energy is supplied to the electron system, this energy is ultimately redistributed among all the carriers by electron-electron scattering, leading to an increase of (collective) electron temperature. Whereas this process depends on several parameters, it typically takes some 10s fs.<sup>[98–100]</sup> Highly excited electrons with the energy exceeding that of the optical phonon energy of  $\approx 0.2$  eV can also relax by interaction with strongly coupled optical phonons (SCOPs), which is also believed to occur on the timescale of some 10s fs.<sup>[100]</sup> For a moderately excited electron system (electron temperature of ≈1000 K), carrier relaxation (i.e., cooling) occurs via coupling to (i) graphene acoustic phonons through disorder-assisted scattering,<sup>[101]</sup> (ii) graphene-optical phonon interactions,<sup>[102]</sup> and (iii) substrate phonons through near-field interactions.<sup>[103,104]</sup> At room temperature, these cooling processes typically take place on a timescale of picoseconds, although longer timescales are predicted to occur in the absence of disorder and a substrate.<sup>[105]</sup>

The nature of the nonlinear response in graphene can be divided into ballistic and diffusive, depending on the timescale of electron transport lifetime (i.e., average momentum scattering time) and the duration of the optical cycles of the driving light wave. Very strong nonlinearity in the electronic motion in graphene in the ballistic regime can be achieved making use of the coherent Bloch oscillations.<sup>[106,107]</sup> They arise from the acceleration of an electron from the Brillouin zone center to the Brillouin zone edge, with  $|\Delta k| = \left|\frac{\pi}{a}\right|$ , in the electric field of the light wave, in a time faster than the average electron momentum scattering time  $\tau = 10-100$  fs, thus making electrons oscillate coherently between  $\Delta k$  and  $-\Delta k$  points in the Brillouin zone (here a = 0.246 nm is the lattice constant of graphene). In the optical or IR ranges, the duration of the optical cycle is on the order of a few femtoseconds, which is faster than the average electron momentum relaxation time in graphene (see Table 1), thus ensuring the ballistic motion of an electron on the timescale of the light wave oscillation. From the acceleration theorem  $\hbar \frac{dk}{dt} = -eE$ , which can be further simplified to  $\hbar \frac{\Delta k}{\tau} = -eE$ , it follows that a minimum electric field on the order of 1–10 MV cm<sup>-1</sup> is

required to reach the regime of coherent Bloch oscillations in graphene with a typical scattering time of  $\tau = 10-100$  fs. Here, *hk* is crystal momentum, *t* is time, *e* is elementary charge, and *E* is electric field strength of the light wave. Such rather high peak fields exceeding 1-10 MV cm<sup>-1</sup> are routinely attainable for optical and IR signals generated with amplified laser systems. Indeed, high-harmonics generation in graphene using coherent Bloch oscillations mechanism was recently demonstrated with IR fields of 10s MV cm<sup>-1</sup> strength by Yoshikawa et al.<sup>[47]</sup> and Higuchi et al.<sup>[108]</sup> We note that coherent Bloch oscillations can be induced at lower field strength in systems with reduced Brillouin zone, such as semiconductor superlattices.<sup>[109,110]</sup>

In contrast to IR or optical light waves, THz waves oscillate on a (sub-)picosecond timescale, considerably longer than the typical electron momentum scattering time in graphene which is on the order of 10-100 fs. Therefore, in the THz range the carrier transport typically occurs in the diffusive regime, where electrons undergo several momentum scattering events within the duration of the THz wave oscillation period (see Table 1), and the microscopic motion of electrons is generally incoherent with the driving electric field. As we will explain in detail, the nonlinearity of graphene related to such diffusive collective motion of free electrons in the applied THz fields is in fact extremely strong, leading to substantial THz saturable absorption<sup>[9,55,57,59]</sup> and extremely efficient THz high-harmonics generation.<sup>[63]</sup> Remarkably, this nonlinearity mechanism only requires THz fields of the order of 10s of kV cm<sup>-1</sup>, which is ≈3 orders of magnitude smaller (i.e., requiring about one million times less peak power) than that required for the optical or IR nonlinearity in graphene via coherent Bloch oscillations.<sup>[47,108]</sup>

In general, THz nonlinearities in graphene can originate from both intraband and interband THz transitions. Simple considerations, however, point out that the THz intraband nonlinear response is expected to be much stronger than the interband one. First, the number of states available for the interband THz transitions is fairly small. Integration of the density of states in graphene (Equation (3)) within 1 THz = 4.1 meV band around the Dirac point yields only about  $N \approx 10^8$  cm<sup>-2</sup> electrons available for interband THz transitions. On the other hand, a typical doping of graphene readily yields a Fermi energy  $E_{\rm F} \approx 100$  meV and free carrier density of  $N_c \approx 10^{12} \text{ cm}^{-2}$  (see Equation (4)),<sup>[59,63,85]</sup> available for the interaction with the THz fields via intraband conductivity, and leading to a typical THz power absorbance on the order of  $A_{intra} \approx 0.1$  at room temperature (see, e.g., refs. [59,63]). Second, as explained in Section 3, the Pauli-blocking of interband THz transitions at elevated temperatures strongly suppresses the interband absorption of graphene, leading to a negligible interaction with the THz fields. The room-temperature power absorbance of graphene at 1 THz frequency is only  $A_{\text{inter}} = \pi \alpha (f_{\text{FD, con-}})$ duction  $-f_{\rm FD, valence}$   $|_{1THz, 300K} \approx 10^{-3}$ , which is  $\approx 100$  times smaller than the typical intraband THz absorbance for doped graphene of  $A_{intra} \approx 0.1$ . Here,  $\alpha$  is the fine-structure constant (see Section 3). Therefore, THz nonlinearities related to interband transitions in graphene can only be observed in (almost perfectly) intrinsic graphene, and only at cryogenic temperatures. Indeed, nonlinear multi-photon-like induced THz absorption in near-intrinsic graphene was recently demonstrated under such conditions, originating from a combination of coherent and incoherent inter- and intraband excitation mechanisms.<sup>[62]</sup> However, this experiment required a multilayer sample (45 layers) in order to significantly enhance the light-matter interaction strength.

At the same time, the intraband transitions in graphene remain active even at room temperature and, as mentioned above, lead to strong THz absorbance with  $A_{intra} \approx 0.1$ . As a result, the interaction of graphene with THz waves via the intraband conductivity mechanism remains much stronger in both linear and nonlinear regimes, as compared to the interband THz transitions.<sup>[9,13,55,59,63,111]</sup>

Figure 4 shows the state-of-the-art of the experimentally measured third-order nonlinear susceptibility  $\chi^{(3)}$  of graphene, in the spectral range from visible to THz. A significant increase





**Figure 4.** Experimentally demonstrated third-order susceptibilities from the NIR to the THz range. Susceptibilities from the following references are plotted over the wavelength of the exciting light pulses: [a] Dremetsika and Kockaert,<sup>[112]</sup> [b] Kumar et al.,<sup>[43]</sup> [c] Soavi et al.,<sup>[46]</sup> [d] Hendry et al.,<sup>[48]</sup> [e] Kundys et al.,<sup>[113]</sup> [f] König-Otto et al. (Landau-quantized graphene in a magnetic field),<sup>[114]</sup> and [g] Hafez et al.<sup>[63]</sup> Whereas in the NIR (left), the nonlinearity mechanism is typically associated with coherent electron dynamics, in the THz (right) the mechanism is noncoherent and thermal. For intermediate wavelengths, different mechanisms have been proposed, such as plasmon nonlocal effects [e], and thermally induced plasmon shifts<sup>[115,116]</sup> (no susceptibility quoted).

of  $\chi^{(3)}$  with the fundamental wavelength of the driving light field is observed, yielding the largest nonlinear coefficient of  $\chi^{(3)} \approx 10^{-9}$  m<sup>2</sup> V<sup>-2</sup> at the THz frequencies (case [g] in Figure 4). This value of  $\chi^{(3)}$  of graphene is several orders of magnitude larger than that of other materials in any spectral range, possibly making graphene the most nonlinear material known to date. We note that even when normalized to material thickness (in case of graphene d = 0.3 nm)<sup>[114]</sup> and using the sheet values for nonlinear coefficients  $\chi^{(3)}_{2D} = \chi^{(3)} d$ , the nonlinearity of graphene still by far exceeds that of other known highly nonlinear systems:  $\chi^{(3)}_{2D} \approx 10^{-19}$  m<sup>3</sup> V<sup>-2</sup> for graphene versus  $\chi^{(3)}_{2D} \approx 10^{-23}$  m<sup>3</sup> V<sup>-2</sup> for highly nonlinear quantum well structures with *d* of a few nm.<sup>[117,118]</sup>

# 4.2. Experimental Observation of THz Nonlinearities in Graphene

In the following, we will highlight the exemplary experimental results and measurement approaches that led to the state-ofthe-art understanding of the THz nonlinearity in graphene.

#### 4.2.1. Experimental Techniques

Various techniques and different sources of intense THz fields have been employed for nonlinear THz spectroscopy of graphene. The most commonly used strong-field THz sources are table-top sources based on optical rectification of amplified laser pulses in various nonlinear crystals,<sup>[55,57–59,62,119,120]</sup> or based on air-plasma THz generation method.<sup>[121]</sup> Most of such experiments represent the nonlinear (variable field strength) version of THz time-domain spectroscopy (THz-TDS), THz pump–THz probe, or THz pump–optical probe experiments, and are performed with spectrally broadband, near single-cycle THz pulses. A comprehensive description of high-field table-top THz sources and nonlinear THz spectroscopic technics can be found in refs. [61,122].

Further, an important role in understanding of the IR and THz optical properties of graphene was played by THz sources at large-scale accelerator-based facilities, such as IR/THz free electron lasers (FELs)<sup>[123]</sup> and superradiant THz undulator facilities,<sup>[124]</sup> providing spectrally dense, narrowband, and broadly tunable output. The most common experiment at FELs has been based on degenerate one-color pump–probe spectroscopy. In a series of careful experimental campaigns, the induced transient differential transmission change in graphene has been measured on Fourier-limited few picosecond timescales over a wide range of frequencies stretching from few THz to mid-infrared frequencies.<sup>[9,114,125,126]</sup> These experiments revealed important insights into the carrier dynamics in graphene, including the pioneering observation of inter- and intraband transitions in doped graphene.<sup>[9]</sup>

Most recently, a prototype superradiant THz undulator facility TELBE<sup>[124]</sup> was used for the pioneering demonstration of THz HHG in single-layer graphene.<sup>[63]</sup> The key for the success of this experiment was a unique combination of emission and detection characteristics of TELBE: (i) spectrally pure, multicycle, phase-stable strong-field pulses of THz and sub-THz radiation delivered at exceptionally high repetition rates, and (ii) novel THz detection schemes based on precise femtosecond timing of the superradiant undulator facility to external laser sources. The former allowed for exceptionally good statistics on the measurement, and the latter enabled a particularly sensitive probing of nonlinear THz signals in the time-domain with subcycle temporal resolution and a dynamic range as high as 10<sup>6</sup>.

#### 4.2.2. Key Observations of Nonlinear THz Absorption in Graphene

Here, we will present the key observations of nonlinear THz absorption in graphene, as well as several results of optical pump–THz probe spectroscopy that have been crucial for the understanding of the nature of THz nonlinearity of graphene.

In 2011, Hwang et al.<sup>[54,55]</sup> reported the pioneering observation of the THz nonlinearity of CVD-grown graphene, using intense single-cycle THz pulses with a maximum fluence of 190  $\mu$ J cm<sup>-2</sup> corresponding to a peak electric field strength of ~100 kV cm<sup>-1</sup>, see **Figure 5**a–d. As experimental methods in this study, nonlinear THz-TDS and THz pump–THz probe spectroscopy were used, as depicted schematically in Figure 5a. In this work, a THz-field-induced transparency of doped graphene, that is, an increase of THz transmission through graphene with increasing the THz pump field, shown in Figure 5b,c, was demonstrated for the first time. The nonlinear transmission enhancement was attributed to saturable absorption effects related to a decrease in the intraband conductivity of the doped graphene sample. The measured power-dependent THz transmission of graphene was described, as shown in Figure 5c, by

1900771 (8 of 25)

ADVANCED SCIENCE NEWS \_\_\_\_\_





**Figure 5.** THz field-induced transparency in monolayer graphene: a) Schematic of varying high-field THz-TDS and intense-THz-pump/weak-THz probe spectroscopy of graphene on a substrate; techniques followed by Hwang et al.<sup>[55]</sup> b) Spectral field transmission increasing with the THz fluence, with the temporal THz fields (left inset) and the corresponding spectra (right inset) transmitted through the graphene sample and the bare substrate, c) The power transmission versus the THz pump fluence; symbols for the experiment and solid red line for the saturable absorption fit, d) Transient differential transmission of the THz probe beam transmitted through THz-pumped graphene as a function of the pump–probe delay time, showing ultrafast (less than 100 fs) transient increase in transmission followed by a subsequent biexponential relaxation over a picosecond time range (b–d: Adapted with permission.<sup>[55]</sup> Copyright 2013, American Chemical Society), and e) suppression of graphene conductivity with THz field for all the THz frequencies in test demonstrating the observed transparency (Reproduced under the terms of the Creative Commons CC-BY 4.0 License.<sup>[59]</sup> Copyright 2015, Springer Nature).

a phenomenological saturable transmission function yielding a saturation fluence of 20  $\mu$ J cm<sup>-2</sup>. An attempt to reproduce the data quasi-quantitatively using a Drude model of the THz intraband conductivity of graphene (see Equation (7)) required variation of the electron momentum scattering time as a free fitting parameter: The trend of reduction of THz conductivity of graphene could be reproduced by reducing the electron scattering time in graphene with increasing the driving THz field.

In the THz pump-THz probe experiment presented in the same work,[54,55] the dynamics of THz absorption bleaching in graphene was demonstrated, with quasi-instantaneous increase in THz transmission followed by a biexponential relaxation on  $\approx$  200 fs and  $\approx$ picosecond timescales (see Figure 5d). The pioneering observation of THz nonlinearity of doped graphene by Hwang et al.<sup>[54,55]</sup> was followed by further experiments on nonlinear THz spectroscopy of graphene<sup>[56,58,59,62,119,120]</sup> using table-top intense near single-cycle THz pulse sources, and similar observations of THz-field-induced transparency (THz absorption bleaching) in both CVD<sup>[56,58,59,111]</sup> and epitaxial<sup>[57,119]</sup> graphene were reported. The physical mechanism of THz absorption bleaching was mainly attributed in all these works to nonlinear intraband conductivity modulation (suppression) by the intense THz driving field (see Figure 5e). However, the precise mechanism of THz conductivity suppression remained under debate at the time.

ADVANCED SCIENCE NEWS \_\_\_\_\_ www.advancedsciencenews.com

In 2014, Bowlan et al.<sup>[62]</sup> observed induced absorption and nonlinear spectral modification of near single-cycle broadband strong-field THz pulses in near-intrinsic epitaxial 45-layer graphene sample at a cryogenic temperature, which was assigned to a complex process initiated by nonlinear THz interband transitions in graphene, vanishing with increasing the sample temperature. In a more recent work, four-wave mixing in multilayer (50 layers) epitaxial graphene in strong magnetic field, and driven by intense far-IR field at 19 THz was reported by König-Otto et al.<sup>[114]</sup>

In 2015, Mics et al.<sup>[59]</sup> published a combined experimentaltheoretical study presenting a thermodynamic picture of the THz nonlinearity in graphene. This work largely reconciled the previous observations of THz nonlinearities in graphene, and provided a quantitative description of the mechanism of the strong modulation of intraband conductivity in the THzexcited graphene using common conservation laws. The basic mechanism of THz nonlinearity of graphene, which will be described in detail in Section 4.3, is as follows. The absorption of the THz field by free electrons of graphene leads to quasiinstantaneous increase in the electron temperature, accompanied by a downshift in the chemical potential, which in turn suppresses the intraband conductivity of graphene.[58,62,81] This rather simple thermodynamic model of the THz nonlinearity of graphene by Mics et al.<sup>[59]</sup> is currently believed to be the most accurate one, showing very good agreement with the experimental results on picosecond-timescale THz nonlinear optics of graphene.

With one-color pump–probe spectroscopy using a tunable FEL, in 2011 Winnerl et al.<sup>[9,126]</sup> studied ultrafast carrier dynamics in quasi-intrinsic (lightly doped) multilayer (70 layers) epitaxial graphene grown on SiC at a range of photon energies from 10 meV (≈2.5 THz) to 250 meV (≈62.5 THz), and made the pioneering observation of, and distinction between, the inter- and intraband transitions in graphene. For photon energies larger than twice the Fermi energy (i.e.,  $\hbar \omega > 2E_F$ ), pump-induced occupation of energy states that are  $\frac{1}{2}\hbar\omega$  above

the Dirac point via interband transition leads to an absorption bleaching of the probe beam similar to semiconductors, resulting in a positive differential transmission signal (increase in transmission), as shown in **Figure 6**a–c for the case of  $\hbar \omega$  = 30 meV. Increasing the pump fluence, as expected, results in a larger differential transmission  $\Delta T/T$  signal.

We note here the *k*-space distribution of the photoexcited carriers in graphene is initially anisotropic. This anisotropy was revealed in pump–probe FEL experiments by examining the polarization relation between the pump and probe beams,<sup>[93,126,127]</sup> as shown in Figure 6d,e. It is also observable for a broad range of IR excitation photon energies and even at THz frequencies,<sup>[128]</sup> for as long as a sufficient pump fluence is available, for both inter- or intraband excitations. This anisotropy vanishes via carrier relaxation by phonon emission within sub-picoseconds or on a longer timescale, depending on the energy of the excited state relative to the optical phonon energies in graphene.<sup>[93,126]</sup> Excitations in lightly doped graphene below the optical phonon energy ( $\approx 200 \text{ meV}$ ) are accompanied by a longer preservation of such an anisotropy over picoseconds due to quenching of the optical phonon emission.<sup>[126,127]</sup>

In contrast to the case of high photon energy inducing interband transitions, pumping at lower photon energies (i.e.,  $\hbar\omega < 2E_{\rm F}$ ) in one-color pump-probe FEL experiment leads to interband absorption of the probe beam: negative probe differential transmission  $\Delta T/T_0$  for  $\hbar \omega = 20$  meV in Figure 6a. This is a result of smearing out the carrier distribution in the band by heating the carriers via free-carrier absorption (intraband dynamics) of the pump beam, thus opening channels for the probing photons of the same energy to induce interband transitions, as illustrated in Figure 6c. We note that this latter observation of stronger THz probe absorption in excited graphene seems to be in qualitative disagreement with the observations of THz probe transmission enhancement by Hwang et al.<sup>[55]</sup> shown in Figure 5d. This discrepancy can nevertheless be accounted for by the much lower doping concentration of the 70-layer graphene sample, and the relatively large photon energies  $\hbar\omega = 20 \text{ meV} = 4.9 \text{ THz}$  employed in the study by Winnerl et al. Indeed, according to the explanation in Section 3, for such high photon energies the Pauli-blocking of interband transitions at room temperature becomes relaxed, thus permitting measurable interband THz absorption in a low-doped multilayer sample.

Let us now consider the optical pump–THz probe experiments, which produced a bulk of valuable knowledge, and significantly contributed to the present understanding of the nature of THz nonlinearity of graphene. In the following, we will notice the similarities between the THz response of graphene to strong THz excitation and to optical excitation.

Optical excitation of graphene significantly modifies the electronic occupation of its band structure. Absorption of an optical pulse by graphene leads to the creation of initial electron–hole pairs, which thermalize among all charge carriers within ~100 fs, leading to a transient electron distribution with an elevated temperature, that is, to carrier heating. This ultrafast electron heating in photoexcited doped graphene, that is, the smearing out of electron Fermi–Dirac distribution in the energy–momentum space, was first directly revealed in 2013 in a pioneering time-resolved angular-resolved photoemission spectroscopy (tr-ARPES) experiment by Gierz et al.<sup>[100]</sup> performed with both inter- and intraband IR excitations.

An important excursion must be made here. If the electronic temperature becomes high enough ( $\approx$ 3000 K), this photo-induced carrier heating also leads to a weak (due to impedance

ADVANCED SCIENCE NEWS \_\_\_\_\_ www.advancedsciencenews.com



**Figure 6.** One-color pump–probe spectroscopy of epitaxial multilayer graphene using narrow-band tunable free-electron laser pulses: a) Differential transmission of the probe beam as a function of the pump–probe delay time at various pump fluences for two cases of photon energy;  $\hbar \omega = 30$  meV > 2*E*<sub>F</sub> and  $\hbar \omega = 20$  meV < 2*E*<sub>F</sub> (Reproduced with permission.<sup>[126]</sup> Copyright 2017, Wiley-VCH, while the original content of the figure is published in Ref. [9]), b) model calculations of the corresponding real conductivity at different electron temperatures over a range of photon energies, showing pump-induced absorption of the probe beam when  $\hbar \omega < 2E_F$  and induced transparency when  $\hbar \omega > 2E_F$  (Reproduced with permission.<sup>[126]</sup> Copyright 2017, Wiley-VCH), c) schematic explanation of the revealed responses in (a) and (b) (Reproduced with permission.<sup>[9]</sup> Copyright 2011, American Physical Society), d) schematic of anisotropic distribution of photoexcited carriers accumulated in the direction perpendicular to the polarization direction of the pump beam (Adapted with permission.<sup>[127]</sup> Copyright 2014, American Chemical Society), e) differential transmission signal revealing the anisotropy described in (d) using one-color pump–probe spectroscopy in which parallel and perpendicular polarizations of the pump and probe beams were compared (Reproduced with permission.<sup>[93]</sup> Copyright 2016, American Physical Society).

mismatch with the far-field), very short-living (<100 fs), yet observable, spectrally broad optical emission, essentially representing the blackbody radiation by the photogenerated hot electron population in graphene. This effect was discovered in 2010 by Lui et al.<sup>[130]</sup> in pristine graphene under 800 nm optical excitation. In particular, the high-frequency tail of the blackbody emission, with photon energies exceeding the excitation photon energy of 1.5 eV, was observed with sensitive photodetectors in the range 1.7–3.5 eV.<sup>[130]</sup> This optical emission due to the blackbody radiation is a collective process of the hot carrier electron distribution, and is thus very different from the conventional semiconductor-type radiative electron-hole recombination. For example, this blackbody optical emission, unlike electron-hole recombination in semiconductors, does not lead to a reduction in the density of excited carriers in graphene. The presence of such blackbody emission in photoexcited graphene,<sup>[130]</sup> and the previously discussed time-resolved ARPES measurements showing the ultrafast increase in electronic temperature in photoexcited graphene,<sup>[100]</sup> together provided convincing proof of the occurrence of photoinduced carrier heating in graphene.

For near-intrinsic or lightly doped graphene with  $E_{\rm F}$  of the order of a few tens of meV,<sup>[84,119,120]</sup> ultrafast photoexcitation leads to an increase of the THz-probed graphene conductivity as a result of the increase in free carrier density, and observed as negative differential transmission of the probe THz pulse.<sup>[131,132]</sup> This effect is similar to above-bandgap photoexcitation of semiconductors.

However, if the photoexcited graphene is significantly doped, the situation becomes quite different. For a typical doping level with  $E_{\rm F} > 100 \text{ meV}$ , several groups<sup>[14,15,84,119,120,129,131,132]</sup> observed a reduction of THz probe absorption of doped graphene as a result of photoexcitation. In Figure 7, we show the exemplary results by Tielrooij et al.<sup>[129]</sup> and Jensen et al.<sup>[84]</sup> This negative photoconductivity is in a way counterintuitive, as it shows that ultrafast optical interband excitation in fact makes doped graphene more THz-resistive. In a combined experimental-theoretical study in 2013, Tielrooij et al.[129] assigned this phenomenon to "hot-carrier multiplication," a cascaded electron-electron scattering process, triggered by the energetic photoexcited electrons, which scatter with the background electrons in the Fermi sea and increase their energy. The microscopic theory for such hot-carrier multiplication by multiple electron scattering in graphene was presented by Song and Levitov in the work of Song et al.,<sup>[98]</sup> where it was also demonstrated that the rate of such electron-electron scattering generally largely exceeds the electron-phonon scattering rate. Such a photoinduced hot-carrier multiplication was connected in the work of Tielrooij et al.<sup>[129]</sup> (see its Supporting Information) to very fast electron thermalization and hence to the increase of the electronic temperature in graphene, leading to the reduction of its chemical potential and conductivity. These studies, explaining the negative photoconductivity of doped graphene, became quite important also for the understanding of www.advancedsciencenews.com





**Figure 7.** Optical pump–THz probe spectroscopy. a) Schematic of the experiment (Reproduced with permission.<sup>[129]</sup> Copyright 2013, Springer Nature), b) THz differential transmission of the peak of the THz probe field as a function of the pump–probe delay time for two cases of optical-pump photon energies maintaining the same photoexcited carrier density of  $\approx 10^{11}$  cm<sup>-2</sup>, showing an increase in the THz transmission of graphene after photoexcitation, corresponding to a suppression of the photoconductivity (Adapted with permission.<sup>[129]</sup> Copyright 2013, Springer Nature). c) Crossover from negative to positive photoconductivity in graphene with decreasing free carrier concentration. Negative photoconductivity in doped graphene and positive photoconductivity in near-intrinsic graphene at fixed photoexcitation level. A sample with a back-gate was used in order to control the position of the Fermi level in graphene (Adapted with permission.<sup>[84]</sup> Copyright 2014, American Chemical Society).

THz-induced nonlinearities in graphene. In Section 4.3, we will show that the final result of both intense THz (with an electric field strength exceeding 10 kV cm<sup>-1</sup>, fluence of  $\approx 2 \,\mu J \text{ cm}^{-2}$ ) and optical excitation (usually with a fluence exceeding 1  $\mu J \text{ cm}^{-2}$ ) of doped graphene is, in fact, equivalent—an increase of the temperature of its background electron population, which in turn reduces its intraband, that is, THz conductivity.<sup>[59,63,84,129]</sup>

The doping concentration in graphene was found to play a significant role in defining the THz response of graphene to both optical and intense THz excitations. Razavipour et al.<sup>[111]</sup> reported the effects of the carrier concentration in graphene on the THz field-induced nonlinearity (only absorption bleaching), using gated graphene samples in which the doping concentration was controlled via electrochemical gating, as depicted schematically in **Figure 8**a. It was found that the THz field-induced transparency in graphene (or the THz nonlinearity in general) becomes more pronounced with increasing the doping concentration, as shown in Figure 8b,c. This indicates that the nonlinear THz field effects in graphene are dominated by intraband conductivity mechanisms, so that the higher the doping concentration the larger the possible modulation of the intraband conductivity by the THz field is.

When the free carrier concentration in the sample was controlled by the combination of electrical doping by gating (static doping) and ultrafast optical excitation, the situation became more complex.<sup>[84,119,120,131,132]</sup> Depending on the initial doping density and the THz probe field strength, the THz photoconductivity was found to be either negative (THz absorption bleaching) or positive (enhancement of THz absorption), as shown in Figure 8d–f. In lightly doped graphene with  $E_{\rm F}$  on the order of a few meV, increased THz absorption in photoexcited graphene was noticed when probing by weak THz field (see Figure 8e), similar to the results in ref. [84] shown in Figure 7c, indicating the semiconductor-like THz response of near-intrinsic graphene. With increase in the THz probe field strength, THz absorption bleaching with a crossover to negative THz photoconductivity (positive THz differential transmission) was observed. On the other hand, for strongly doped graphene the THz photoconductivity always remained negative, indicative of THz absorption bleaching (positive THz differential transmission) as a result of combined action of THz and optical excitation, as shown in Figure 8f.

In a more recent work, Tomadin et al.<sup>[133]</sup> significantly extended the understanding of the effect of photoexcitation on graphene based on a combination of microscopic modeling and optical pump–THz probe spectroscopy. Interestingly, it was found that depending on the Fermi energy of the sample, the ultrafast electronic thermalization in photoexcited graphene can lead







**Figure 8.** Carrier density dependence of the THz field-induced nonlinearity in graphene: a) Schematic of the interaction of intense THz field with Fermi-level-tunable graphene, b,c) nonlinear THz transmission of gated graphene as a function of both the THz field strength and the doping level expressed by the gating voltage relative to the voltage corresponding to the charge neutrality point  $V-V_{CNP}$  (Reproduced with permission.<sup>[111]</sup> Copyright 2015, American Physical Society). d) Schematic of optical pump-induced dynamics probed by intense THz field, and e,f) peak of THz differential transmission signal  $\Delta T/T_0|_{peak}$  of photoexcited graphene as a function of the THz probe field for lightly doped graphene pumped by 29 µJ cm<sup>-2</sup> optical pump fluence (Adapted with permission.<sup>[120]</sup> Copyright 2015, AIP Publishing), highly doped graphene at three optical pump fluences of 29, 79, and 137 µJ cm<sup>-2</sup> (Adapted with permission.<sup>[119]</sup> Copyright 2015, American Physical Society).

not only to "hot-carrier multiplication," that is, electron heating with the total number of free carriers conserved (case of larger  $E_{\rm F} > 0.1$  eV), but under certain conditions also to a "real" free carrier multiplication via interaction with the valence band states (case of smaller  $E_{\rm F} < 0.1$  eV). Carrier multiplication in graphene was first predicted by Winzer et al.<sup>[134]</sup> and observed experimentally by Plötzing et al.<sup>[135]</sup> Ref. [133] furthermore provides insights into the mechanism leading to the reduction of the graphene conductivity with increasing electron temperature, based purely on electron kinetics effects, including the effect of screening, and considering that momentum scattering is dominated by long-range Coulomb scattering with impurities outside graphene.

An interesting observation was made by Tani et al.<sup>[121]</sup> in a THz pump-optical probe experiment, performed using  $\approx 30$  fs single-cycle THz pump pulse and NIR probe of 800 nm wavelength (1.55 eV photon energy). A standard doped CVD-grown graphene was used in this work. It was shown that the excitation of graphene with the THz pulses with electric field of up to 300 kV cm<sup>-1</sup> resulted in an ultrafast (≈150 fs) transient increase of the optical probe transmission through graphene, as shown in Figure 9. This effect was assigned to absorption bleaching of the optical probe by Pauli-blocking of the NIR interband transitions at 1.55 eV photon energy under intense ultrafast THz excitation of graphene. The authors attributed the electron occupation of states around 0.775 eV above the Dirac point to impact-ionization effects. The impact ionization is the process of electron-hole pair generation resulting from the loss of kinetic energy of another highly energetic electron or hole, which was assumed to be ballistically accelerated by the strong pump THz field. It was suggested that this process leads to carrier multiplication, allowing for the occupation of higher energy states around 0.775 eV above the Dirac point, in turn bleaching the interband optical absorption of the 1.5 eV probe pulse in graphene. The impact ionization in strong (up to 300 kV cm<sup>-1</sup>), ultrafast (~30 fs) THz field pulses as observed by Tani et al.<sup>[121]</sup> increases the free carrier density in graphene. This is an additional effect to the THz-induced electron heating—the main effect of interaction of doped graphene with intense THz fields, which as such conserves the free carrier density (see Section 4.3 for details). To the best of our knowledge, the impact ionization leading to carrier multiplication has not been observed in the experiments dealing with "slower" (>1 ps long) strong-field THz pulses as used in the experiments of refs. [56–59,63,111,119].

Besides standard "continuous" graphene samples, nonlinear THz spectroscopy was also performed on patterned graphene, featuring the confinement of THz field, and thus leading to enhancement of the THz nonlinearity. For example, it was theoretically predicted<sup>[136-138]</sup> and experimentally veri- $\operatorname{fied}^{[51,115,116]}$  that enhancement of the nonlinear interaction of the light field in graphene can be achieved by local field confinement via resonant plasmonic absorption if the graphene is patterned in subwavelength structures such as ribbons. Using THz FEL, Jadidi et al.[115,116] demonstrated nonlinear THz absorption at plasmonic resonance in lithographically patterned epitaxial graphene ribbons (see Figure 10a), following one-color pump-probe scheme with THz field polarization perpendicular to the patterning direction. The plasmonic resonance frequency (see Figure 10b) was defined by the dimensions of the graphene pattern, as well as by the electron density and temperature. The THz pump at resonance resulted in two consequent effects: (i) a rise in the electron temperature by the THz excitation enhanced due to the THz field confinement, and consequently (ii) a redshift in the resonance absorption. The probe experienced absorption







**Figure 9.** THz pump/optical probe of single layer graphene: a) Schematic demonstration of the THz field induced dynamics in hole-doped CVD graphene, showing ballistic acceleration imposing impact ionization from the valence band to the conduction band leading to occupation of states that block further interband transitions by the optical probe beam, b) the THz pump field with a peak field of 300 kV cm<sup>-1</sup> with the corresponding spectrum shown in the inset, c) negative differential optical density corresponding to enhanced transmission of the optical probe beam as a function of the pump–probe delay time, d,e) experimental data (symbols) of the optical density change due to the intense THz pump as a function of the pump–probe delay time, and model calculations (solid lines) based on the Boltzmann equation with scattering mechanisms including impact ionization and Auger recombination for THz field strengths of 170 and 300 kV cm<sup>-1</sup>, respectively, f,g) calculated carrier density following the impact ionization and Auger a scaling with  $E_{TH_2}^{2.5}$  and a better fit with the model only when impact ionization and Auger recombination were considered (Reproduced with permission.<sup>[121]</sup> Copyright 2012, American Physical Society).



**Figure 10.** Plasmonic-induced enhancement of THz nonlinearity in lithographically patterned graphene ribbons: a) Schematic of lithographically patterned graphene ribbons on SiC substrate with a ribbon periodicity *L* and a ribbon width *W*. b) Broadband THz transmission through the sample showing plasmonic absorption resonance at 3.8 THz (the dark cyan line with arrow indicating the relevant axis). The normalized spectrum in red corresponds to one of the FEL pulses (here at resonance) employed in the degenerate pump–probe spectroscopy (with arrow indicating relevant axis) (Replotted with data provided by the authors and with permission.<sup>[116]</sup> Copyright 2019, American Chemical Society). c) THz differential transmission at resonance of the probe beam as a function of the pump–probe delay time for various THz pump fluences (Reproduced with permission.<sup>[116]</sup> Copyright 2019, American Chemical Society), e) the differential transmission signal as a function of the pump fluence; colored symbols for the experiment and a solid black line for the  $\sqrt{F}$  fit (Reproduced with permission.<sup>[116]</sup> Copyright 2019, American Chemical Society), e) the differential transmission signal at various frequencies around the resonance frequency showing a sign reversal when the probe frequency is tuned to be below resonance (Reproduced with permission.<sup>[116]</sup> Copyright 2019, American Chemical signal in (e) as a function of frequency (Reproduced with permission.<sup>[116]</sup> Copyright 2019, American Chemical Society).

1900771 (14 of 25)





bleaching, with positive differential transmission signal  $\Delta T/T_0$  increasing as the square root of the pump fluence, as shown in Figure 10c,d. The relatively small pump fluence employed here, as compared to that employed in the studies with continuous planar graphene,<sup>[9,55,59]</sup> showed a significant enhancement in the THz field-induced nonlinearity by resonant plasmons in patterned graphene. On the other hand, the redshift of the plasmon resonance due to the rise in the electron temperature induced by the THz pump led to increased absorption of the probe beam when probing at frequencies lower than the original resonance frequency (negative  $\Delta T/T_0$  in Figure 10e,f).

# 4.3. The Thermodynamic Model of Nonlinear Intraband THz Response of Graphene

The observations of the nonlinear THz response of graphene, described in Section 4.2.2, can be summarized as follows. For doped graphene, excitation with intense THz fields (field strength above  $\approx 10 \text{ kV cm}^{-1}$ , fluence of the order of  $\approx \mu J \text{ cm}^{-2}$ ) leads to a

reduction of the THz conductivity of graphene. The effect of THz excitation of doped graphene is thus the same as the effect of optical excitation (also summarized in Section 4.2.2), which similarly reduces the THz, that is, intraband, conductivity of graphene.

In the following, we will describe the basic mechanism of such a THz-driven conductivity suppression, following the thermodynamic model of ultrafast charge transport in graphene presented by Mics et al. in 2015.<sup>[59]</sup> This model, resting on basic conservation laws, self-consistently treats the THz-induced electron heating in doped graphene during its interaction with the incident THz field, dynamically modifying the propagation conditions for the THz field itself. The key physical processes described by the thermodynamic model of THz nonlinearity of graphene are summarized in **Figure 11**.

We note here that most equations of this model are only strictly valid in the Sommerfeld expansion regime, that is, at electron temperature being below the Fermi temperature:  $T_e < T_F$ , where  $T_F = E_F/k_B$ . For highly doped graphene on SiO<sub>2</sub> with typical Fermi energy of 0.25–0.4 eV, the Fermi temperature is  $T_F = 3000-4800$  K. For practical purposes, however, the approximation  $T_e < T_F$  in the description of nonlinear interaction of



**Figure 11.** The thermodynamic model of THz-driven electron transport in graphene: a) Schematic for the THz field-induced electron dynamics via intraband THz conductivity (THz absorption by the Fermi surface electrons) showing intraband heating enabled by *e–e* scattering that allows electrons in the band to thermalize among themselves, which is accompanied by a reduction of the chemical potential to keep the carrier density constant (Reproduced under the terms of the Creative Commons CC-BY 4.0 License.<sup>[59]</sup> Copyright 2015, Springer Nature), b) the population density before (blue) and after (red) the THz excitation (Reproduced with permission.<sup>[63]</sup> Copyright 2018, Springer Nature), c) the corresponding Fermi–Dirac distribution before and after the THz excitation showing a smearing-out of the hot carrier distribution in the band accompanied by a downshift in the chemical potential to maintain the total carrier density (the area under the curves in (b)) constant, and d) illustration of spectral weight conservation for inter- and intraband transitions in graphene. Reduction of chemical potential for hotter electron population leads to increase of the spectral weight for interband transitions (as a result of Pauli-unblocking), which is compensated by the reduction of the spectral weight of intraband conductivity in graphene (c,d: Reproduced with permission.<sup>[82]</sup> Copyright 2015, IOP Publishing).

arbitrary THz field with graphene remains reasonably accurate. Most of the nonlinear response in graphene is in fact achieved in the regime of  $T_{\rm e} < T_{\rm F}$ , and the effect of further increase in electron temperature appears smaller in comparison. For higher electron temperatures, leading to chemical potential approaching the Dirac point, the accuracy of the thermodynamic model by Mics et al.<sup>[59]</sup> can be improved by including the valence band states, as shown by Frenzel et al.<sup>[132]</sup>

Let us first consider the energy conservation in the interaction of THz radiation with doped graphene. The application of the THz field  $E_{\text{THz}}$  to graphene leads to generation of THz currents according to Ohm's law,  $j = \sigma_{\text{intra}} E_{\text{THz}}$ , where  $\sigma_{\text{intra}}$  is the intraband conductivity of graphene, described by Equation (6). Generally, the conductivity is directly proportional to the power absorption coefficient of the material  $\alpha = \sigma'/cn\varepsilon_0$ , where  $\sigma'$  is the real part of conductivity (see Equation (6)), n is the real part of refractive index, c is speed of light in vacuum, and  $\varepsilon_0$  is vacuum permittivity. The excitation of THz current j in doped graphene thus facilitates the transfer of energy from the absorbed THz field to the free conductive electrons in graphene. In the approximation of sharp Fermi edge, which is convenient for this discussion, one can assume that this energy is initially confined to the electrically conductive states residing at the Fermi level  $E_{\rm F}$ .

The linear energy-momentum dispersion in graphene given by Equation (2), apart from the high electron mobility due to fundamental absence of Coulomb binding, yields yet another unique property: perfect energy and momentum conservation for collinear electron-electron scattering. In a simple intuitive picture, given available phase space, any two Dirac electrons in graphene can scatter by conserving their total energy and momentum: One electron increases its energy and momentum by  $\Delta E$  and  $\Delta k$ , respectively, while the second electron decreases its energy and momentum by the same amounts (see Figure 11a). Detailed considerations indeed showed that collinear scattering between two graphene electrons is the most probable, and therefore the fastest electron scattering process.<sup>[98,99,139]</sup> This is in contrast to materials with quasi-parabolic energy-momentum dispersion, such as semiconductors, where typically more than just two participating particles are needed in order to fully conserve both energy and momentum in a scattering event, thus reducing its probability and consequently slowing the scattering rate down.

It is via these collinear electron–electron interactions in graphene, that the energy of the absorbed THz field, initially confined to the conductive states at the Fermi edge, is being redistributed across the entire carrier population of graphene, becoming its collective kinetic energy—the electronic heat. The process of such an energy redistribution, the electron thermalization, is known to happen on an ultrafast timescale of <100 fs for both intra- and interband excitation of doped graphene.<sup>[100,140–143]</sup> This thermalization is much faster than the oscillation cycle of THz waves, which is on the order of picoseconds. As a result, one can treat the thermalization dynamics as an instantaneous process on the timescale of THz–graphene interaction, which leads to the following thesis:

(1) Energy conservation: The THz energy absorbed by graphene electrons via intraband conductivity is instantaneously converted into electronic heat with perfect conversion efficiency, raising the electronic temperature. Since the electrons are assumed as internally thermalized at all times, and hence are described at all times by Fermi–Dirac distribution  $f_{\rm FD}(\varepsilon,\mu,T_{\rm e})=1/\left[\exp\left(\frac{\varepsilon-\mu}{k_{\rm B}T_{\rm e}}\right)+1\right]$  with a certain temperature  $T_{\rm e}$  and chemical potential  $\mu$ , the free electron conductivity of a THz-driven graphene at all times can be described by Equation (6) depending on the above parameters. Quantitatively, the total thermal energy density (energy per unit area) of the graphene electronic system excited by THz field can be expressed as  $Q_0 + \Delta Q = \int_0^{\infty} \varepsilon D(\varepsilon) f_{\rm FD}(\varepsilon,\mu,T_{\rm e}) d\varepsilon$ , where  $D(\varepsilon) = 2|\varepsilon|/\pi(\hbar\nu_{\rm F})^2$  is the density of states. Here,  $Q_0$  is the electronic thermal energy before the THz field excitation, that is, with the chemical potential  $\mu$  equals to the initial Fermi energy at the initial temperature (e.g., room temperature). With the THz field absorption, an additional heat  $\Delta Q$  is added, thus leading to a rise in  $T_{\rm e}$ , as illustrated in Figure 11a.

Second, we need to impose a condition on electron density conservation for the intraband THz excitation of graphene: at any electron temperature  $N_c = \int_0^{\infty} \varepsilon D(\varepsilon) f_{FD}(\varepsilon, \mu, T_c) d\varepsilon = \text{constant}$ . This requires that the rise in temperature  $T_e$  due to the THz deposited heat  $\Delta Q$  must be accompanied by a reduction in the chemical potential  $\mu$ , as shown in Figure 11b,c (see also ref. [82]). The condition of carrier density conservation thus leads to the second thesis of the thermodynamic model:

(2) Carrier density conservation: Increase of electron temperature leads to reduction of chemical potential in THz-excited graphene.

Now that we established that the intraband THz absorption in graphene leads to an increase of electron temperature accompanied by a downshift in the chemical potential, we arrive at a qualitative understanding of the reduction of THz conductivity in THz-excited graphene, based on the spectral weight conservation rule, illustrated in Figure 11d. This rule was first applied to the treatment of THz photoconductivity in graphene by Frenzel et al.<sup>[132]</sup> Indeed, the increase of the electron temperature, resulting in the downshift of the chemical potential  $\mu$ , leads to unblocking of spectral density for previously Pauli-blocked interband transitions at the (infrared) photon energy of  $\hbar \omega > |2\mu|$ . This increases the spectral weight for interband transitions for graphene with hotter electron population, whereas the total spectral weight of both interband and intraband transitions must remain conserved. This brings us to the third thesis of the thermodynamic model.

(3) Spectral weight conservation: The downshift of chemical potential in graphene with higher electron temperature leads to an increase of the spectral weight for interband transitions, compensated by a decrease of the spectral weight for intraband transitions. The latter reduces the THz conductivity in graphene with higher electron temperature.

Now we consider the dynamics of the processes underlying the thermodynamic model of conduction in graphene. While the energy transfer from the driving THz field to the free electron population of graphene, raising the electronic temperature, occurs quasi-instantaneously on sub-100 fs timescale, the consecutive electron cooling occurs on a slower, few-picosecond timescale of phonon emission, as schematically depicted in Figure 11b. This dynamics of electron cooling after intraband excitation of doped graphene was measured in the pioneering



tr-ARPES experiments by Gierz et al.,<sup>[100]</sup> see **Figure 12a**. The electron cooling dynamics features three distinct timescales ranging from  $\approx 10$  fs to 1 ps, which are assigned to the electron relaxation via optical and acoustic phonons, respectively (see also Table 1). The disparity between electron heating (sub-100 fs) and total cooling (several ps) timescales naturally leads to the heat accumulation in the electronic population of THz-driven graphene, ultimately resulting in the reduction of THz conduc-

tivity in graphene on the timescale of the oscillation cycle of the THz field. Note how this dynamics of electron cooling, directly observed in tr-ARPES measurements,<sup>[100]</sup> roughly corresponds to the dynamics of THz conductivity recovery in a THz pump–THz probe experiments by Hwang et al.<sup>[54,55]</sup> (see Figure 5d).

Further, the tr-ARPES data by Gierz et al.<sup>[100]</sup> demonstrated saturation of the electron temperature in graphene with increasing intraband excitation fluence, see Figure 12b. Such



**Figure 12.** Dynamics of electron heating and cooling under intraband excitation, and its effect on the THz conductivity and absorbance of graphene: a) Dynamics of electron heating and cooling in doped graphene under intraband excitation as measured in tr-ARPES experiment. b) Saturation of electronic temperature with increasing intraband excitation fluence (a,b: Reproduced with permission.<sup>[100]</sup> Copyright 2013, Springer Nature). c) Temporal evolution of electronic temperature in graphene with  $E_F = 70$  meV under quasi-single-cycle THz excitation of variable field strength (gray area indicates the modulus of driving THz field). d) Measured (symbols) and calculated (solid lines) nonlinear conductivity spectra of doped graphene, as the peak THz field increases from 2.3 kV cm<sup>-1</sup> (yellow) to 120 kV cm<sup>-1</sup> (black). e) Measured and calculated nonlinear absorption of doped graphene with  $E_F = 70$  meV, and the accumulated thermal energy density retained in the electronic temperatures as functions of THz peak field strength (arrows indicate relevant axis) (c–e: Reproduced under the terms of the Creative Commons CC-BY 4.0 License.<sup>[59]</sup> Copyright 2015, Springer Nature).

a saturation indicates a fundamental nonlinearity in the intraband excitation of graphene: raising the electron temperature suppresses the intraband conductivity, thus limiting the energy transfer rate for the following portion of the excitation signal. This saturation occurs on top of the sublinear relation between the electron temperature and the incident power, which is caused by the electron heat capacity increasing with electron temperature. Note again the similarity in the saturation of tr-ARPES—measured electronic temperature in the work by Gierz et al.<sup>[100]</sup> (Figure 12b) and the saturable THz absorption in THz doped graphene from the pioneering work of Hwang et al.<sup>[54,55]</sup> (Figure 5c).

ADVANCED SCIENCE NEWS \_\_\_\_\_ www.advancedsciencenews.com

Now that both the mechanism of energy transfer from THz field to graphene— the intraband conductivity according to Equation (6), and its dynamics—quasi-instantaneous electron heating and picosecond-timescale electron cooling (see Figures 11b and 12a) are established, one can numerically model the interaction of graphene with the driving THz fields in the time domain.

We note here that the intraband conductivity given by Equation (6) also contains an electron energy-dependent scattering time  $\tau(\varepsilon)$ . For graphene in full thermal equilibrium, that is, before strong THz excitation, this scattering time corresponds to the scattering time at the Fermi level  $\tau(E_{\rm F})$ , which can be determined experimentally in a linear THz-TDS experiment such as Figure 2f. Even though several electron scattering regimes are postulated for (hot) electrons in graphene, there is no full consensus on this topic in the literature until now. The two dominating momentum scattering mechanisms are known to be the long-range Coulomb scattering and short-range disorder scattering, for which the scattering times can be approximated as, respectively,  $\tau(\varepsilon) = \gamma \varepsilon$  and  $\tau(\varepsilon) = \beta/\varepsilon$ , where  $\gamma$  and  $\beta$  are linear proportionality constants (see, e.g., refs. [28,59] for discussion). Typically, long-range Coulomb scattering with  $\tau(\varepsilon) = \gamma \varepsilon$  is believed to dominate for CVD-grown graphene deposited on a regular dielectric substrate (such as SiO<sub>2</sub>), whereas this scattering channel can be suppressed in exfoliated graphene that is suspended or encapsulated by hexagonal boron nitride. Assuming that one of these scattering mechanisms dominates, the proportionality constant, whether  $\gamma$  or  $\beta$ , can thus be established experimentally from a linear THz-TDS experiment at room temperature as  $\gamma = \tau(E_{\rm F})/E_{\rm F}$  and  $\beta = \tau(E_{\rm F})E_{\rm F}$ , respectively. Both the single value for scattering time  $\tau$ , and the Fermi energy  $E_{\rm F}$  can be readily determined in such a measurement (see Figure 2c-f and refs. [59,63,144]).

Once the dominating scattering mechanism for hot electrons, and hence the energy dependence of the scattering time, is assumed, one directly applies the conditions of energy and electron density conservation to the interaction of the driving THz field with graphene. The experimentally measured incident THz field is then numerically propagated in the time domain through the graphene, which at each point in time is characterized by instantaneous conductivity from Equation (6) dependent on instantaneous value of electron temperature  $T_e$  and chemical potential  $\mu$ , and with the energy-dependent scattering time  $\tau(\varepsilon)$ . This numerical propagation generates the transmitted THz field that can be analyzed in the same fashion as in nonlinear THz-TDS experiments, yielding the nonlinear, field-dependent conductivity spectra of graphene.

Such a numerical propagation can be performed fully numerically as is customary in traditional nonlinear optics (see, e.g., refs. [145,146] and references therein), or using convenient semi-analytical parameterization (see refs. [59,63] including the Supporting Information and Methods sections).

In Figure 12c,d, the results of such a numerical propagation of the experimentally measured incident THz pulse through the CVD graphene with  $E_{\rm F} = 70$  meV and  $\tau(E_{\rm F}) = 140$  fs are shown, assuming the long-range Coulomb scattering scenario for the hot electrons.<sup>[59]</sup> Figure 12c shows the calculated evolution of the electronic temperature in graphene, driven by near-single-cycle THz pulse of variable field strength, whose modulus is shown as a gray area. The evolution of electronic temperature  $T_{e}$  generally reproduces the dynamics of the THz field. However, on the timescale of the light-matter interaction  $T_{\rm e}$  never reaches the initial temperature of 300 K, since the total cool-off of the electronic population is a few-picosecond timescale process, which is only complete after the main part of the driving THz field has gone (see ref. [100] and Figure 12a). This results in permanent electronic heat accumulation in graphene on the timescale of the interaction with the driving THz field, reducing the overall THz conductivity and absorbance of graphene, as the THz field is transmitted through it.

The numerical transmission of the THz field through graphene using the thermodynamic model yields the nonlinear THz spectra of graphene, shown in Figure 12d (solid lines) together with the experimental data (symbols), for the THz fields ranging from 2.3 to 120 kV cm<sup>-1</sup>.<sup>[59]</sup> The conductivity in THz-driven graphene decreases with both the THz field strength and the frequency. It is evident that the agreement between the calculations and the entirety of the experimental data is quantitative, lending credence to the theses underlying the thermodynamic model of the nonlinear THz conduction of graphene.<sup>[59]</sup>

Figure 12e shows the measured and the calculated power absorption of the graphene sample as function of the THz field strength, demonstrating a very strong saturable absorption effect, as has also been observed in many previous measurements (see, e.g., refs. [55,56,58,59,111] and Section 4.2 for discussion). The THz power absorption of doped graphene, measured here over the full timescale of THz-graphene interaction of several picoseconds, reduces from over A = 0.125down to almost zero as the driving THz field increases from 2.3 to 120 kV cm<sup>-1</sup>. This effect of very strong THz intraband absorption bleaching is in perfect agreement with the saturation of electron temperature in doped graphene under intraband excitation, observed in tr-ARPES experiments,<sup>[100]</sup> which, as mentioned before, in turn limits the energy transfer rate from the driving THz field to graphene electrons. In the same figure, the calculated accumulated electron heat, as well as the peak and average (in brackets) electronic temperature of graphene is presented. The energy balance shows that in this case about 15% of the absorbed THz energy remained within the electronic system of graphene on the timescale of interaction with the THz field, irrespective of the THz field strength. The calculated electronic temperatures in 1000s of Kelvin range, reached as a result of THz excitation, are in agreement with the measurements of electronic temperature in tr-ARPES experiments.[100]

As we already mentioned in Section 4.2.2, the thermodynamic model of intraband electron dynamics in graphene is also applicable to the THz response of doped graphene under optical interband excitation. The tr-ARPES experiments with interband pump,<sup>[100,147,148]</sup> as well as optical pump-probe spectroscopy<sup>[139,149]</sup> showed a similar effect of increase of electron temperature in doped graphene subject to interband excitation. Indeed, also in the case of interband optical excitation the collinear electron-electron scattering of Dirac electrons will lead to at least partial redistribution of the energy of photoexcited electron-hole pairs to the background free carrier population, thus increasing its temperature and driving down THz conductivity of graphene.<sup>[84,129,133]</sup> As a result, THz conductivity of photoexcited graphene allows one to estimate the electronic temperature, and hence the efficiency of electron heating induced by optical excitation. We note here that the electron heating efficiency under optical interband excitation is not supposed to be perfect, unlike the THz-driven intraband heating efficiency. The reason is that the photoexcited electrons and holes usually possess enough energy in order to emit optical phonons of ≈200 meV energy, and hence to transfer their energy directly to the lattice of graphene, without exciting the background electron population. Optical phonon emission thus becomes the competing process to the background electron heating in graphene. In ref. [84], Jensen et al. used optical pump-THz probe spectroscopy to estimate the optically driven electron heating efficiency in photoexcited doped graphene. Using a numerical version of the model by Song and Levitov, ref. [129] and its Supporting Information, the photoconductivity spectra of photoexcited graphene were described, thus extracting the dependence of the electronic temperature and the total thermal energy of the electron gas in graphene on the fluence of photoexcitation. It was shown that in the case when the total energy of the unexcited electron gas ("depth of the Fermi sea") was significantly lower than the absorbed energy of the interband photoexcitation of graphene, all the energy of the photoexcited electrons and holes was converted into electron heat, without the direct excitation of the lattice via optical phonon emission. However, for higher excitation fluence and lower Fermi energy, the electron heating efficiency was found to decrease at the expense of the direct coupling of photoexcited electrons to the lattice. Tomadin et al.<sup>[133]</sup> further developed the findings of the work of ref. [84] using a more advanced microscopic model.

#### 4.4. THz High-Harmonic Generation

As described in the previous section, the interaction of strong THz field E(t) with the free electrons in graphene results in strong suppression of graphene conductivity via carrier heating on sub-100 fs timescale, which is followed by the conductivity recovery via carrier cooling by phonon emission, occurring on the timescale of a few picoseconds (see Figure 11). As a result, during the interaction with the THz field, the instantaneous conductivity of graphene  $\sigma_{intra}(E,t)$  becomes time- and field-strength dependent. This conductivity nonlinearity in turn leads to the generation of nonlinear current in graphene  $j(t) = \sigma_{intra}(E,t)E(t)$  in response to the driving THz field E(t). If the driving THz field is monochromatic, that is, it oscillates

at a defined frequency *f*, the induced nonlinear current in graphene will contain oscillations at higher order harmonics of the driving frequency. Due to the centrosymmetry of graphene, even-order harmonic generation is forbidden, and thus only odd-order higher harmonics 3f, 5f, 7f, ... etc., will be generated. We note that the generation of high-order THz harmonics in graphene has long been proposed in theory (see, e.g., refs. [19,22,23]), yet the experimental demonstration of this long-sought effect has been achieved only recently.

In 2018, Hafez et al.<sup>[63]</sup> have demonstrated extremely efficient THz high-harmonics generation in graphene using the thermodynamic response of its free electrons to driving quasi-monochromatic THz fields, following the thermodynamic model by Mics et al.<sup>[59]</sup> The multicycle quasi-monochromatic THz fields with peak electric field up to 85 kV cm<sup>-1</sup> at frequencies between 300 and 700 GHz, used as a pump signal, were generated by a superradiant THz undulator facility TELBE at Helmholtz-Zentrum Dresden-Rossendorf.<sup>[124]</sup> As a sample, a regular CVDgrown graphene deposited on a SiO<sub>2</sub> substrate was used, with the Fermi energy of  $E_{\rm F} = 170$  meV, and the electron momentum scattering time at the Fermi level of  $\tau(E_{\rm F}) = 47$  fs.

The quasi-monochromatic multicycle THz driving of graphene significantly enhances its effective nonlinear conductivity response, as compared to the single-cycle excitation demonstrated in refs. [55,56,58,59,111]. The key reason is that the electron cooling time, which is on the order of picoseconds, conveniently matches the oscillation period of a multicycle pump THz wave. This allows the graphene conductivity (and hence its absorbance) to efficiently recover in time for the next electric field cycle to arrive. Such an efficient recovery of conductivity of graphene, in turn, facilitates more efficient energy transfer from the THz pump wave to graphene electrons over the whole timescale of THz–graphene interaction, which for a multicycle THz pump wave can last 10s of picoseconds.

Figure 13 shows the thermodynamic calculations, following the model by Mics et al.<sup>[59]</sup> of the THz-induced electron heat and the corresponding temporal modulation of graphene absorbance (and hence its conductivity), driven by the multicycle THz field oscillating at a fundamental frequency of 0.3 THz. This calculation particularly exemplifies the highly nonlinear nature of response of free electrons in graphene to THz excitation: the driving field strengths of 10 and 85 kV cm<sup>-1</sup> lead to dramatically different effects on the electron temperature in graphene, and its absorbance. If the effect of 10 kV cm<sup>-1</sup> excitation is minor, leading only to a slight perturbation of the equilibrium properties of graphene, the excitation with 85 kV/ cm<sup>-1</sup> THz field strength results in a large THz-induced heat deposition into the system, and significant absorption and hence conductivity modulation featuring pronounced temporal oscillations observable twice per optical cycle of the driving field. This is a particular consequence of the centrosymmetry of graphene, where both the positive and the negative THz field oscillation half-cycle produces the same effect on its properties, in this case, suppressing its conductivity.

**Figure 14** shows the schematic of the experiment from the work of Hafez et al.,<sup>[63]</sup> the spectrum of the driving THz field at the frequency f = 0.3 THz, as well as the spectrum of the THz field transmitted through graphene sample, with clearly visible higher odd-order THz harmonics up to the seventh order. The





**Figure 13.** THz field-induced heating and absorption modulation: a) Multicycle quasi-monochromatic THz driving field oscillating at 0.3 THz, b,c) the temporal THz induced electronic heat density (energy per unit area), and the THz power absorbance, respectively, of graphene driven by the intense multicycle THz pulse in (a) at two THz peak field levels of 10 kV cm<sup>-1</sup> (the blue curves) and 85 kV cm<sup>-1</sup> (the red curves), demonstrating the thermal response of the Dirac electrons of doped graphene to the intense THz driving field (Reproduced with permission.<sup>[63]</sup> Copyright 2018, Springer Nature).

fundamental field at f = 0.3 THz is transmitted through the graphene with the transmission coefficient exceeding 0.9, which serves as a convenient reference for the conversion efficiency from the THz pump field to higher harmonics. These conversion efficiencies are remarkably high, especially for a material which is

only one atomic layer thick: They demonstrate the values slightly less than 10<sup>-2</sup>, 10<sup>-3</sup>, and 10<sup>-4</sup>, for the third, fifth, and seventh THz harmonics, respectively. The total nonlinear field conversion efficiency therefore tends to  $\approx$ 1%. The spectral range in this experiment was limited by the detector bandwidth, however, the signal-to-noise ratio permitted by the TELBE THz source and the detection scheme<sup>[124]</sup> was not a limiting factor in this measurement.

In Figure 15a,b, the measured HHG signals in the time domain, as well as the corresponding results of the thermodynamic model calculations using the experimental pump wave are presented. The agreement between the measurement and the calculation is rather good: The thermodynamic calculation captures well all the key features of experimental signals, including their magnitude, distortion, and the shortening of the generated harmonic pulses with increase in their order. Figure 15c summarizes the experimental result of THz HHG in graphene by showing the electric field strength of the generated third, fifth, and seventh harmonics, as a function of the pump field strength (symbols). In the same figure, the results of the thermodynamic model calculation are shown, using the experimental pump field and basic parameters of graphene in full thermal equilibrium at 300 K as an input. The agreement between the data and the model is quantitative. From the observed dependencies in the regime of small-signal nonlinearities (pure power law, dashed lines), one can now estimate the effective nonlinear optical coefficients of graphene up to the seventh order:  $\chi^{(3)}_{\text{eff}} \approx 1.7 \times 10^{-9} \text{ m}^2 \text{ V}^{-2}$ ,  $\chi^{(5)}_{\text{eff}} \approx 1.2 \times 10^{-22} \text{ m}^4 \text{ V}^{-4}$ , and  $\chi^{(7)}_{eff} \approx 1.74 \times 10^{-38} \text{ m}^6 \text{ V}^{-6}$  (see ref. [63] and its Methods section for details). These values are extremely large, and by many orders of magnitude exceed the nonlinear coefficients of any other known electronic material. This possibly makes graphene the most nonlinear material known to date.

Now that the thermodynamic model of THz response of graphene has demonstrated its high accuracy in reproducing



**Figure 14.** THz high-harmonic generation in single-layer graphene up to the seventh order at room temperature: a) schematic of the THz odd-order harmonics re-emitted from graphene driven by a multicycle THz field, b) experimental data of the THz field spectrum transmitted through the graphene sample (in blue) showing odd harmonics up to the seventh order relative to the spectrum of the driving field at a fundamental frequency of 0.3 THz (in red) (Reproduced with permission.<sup>[63]</sup> Copyright 2018, Springer Nature).





**Figure 15.** THz high-harmonic generation in single-layer graphene up to the seventh order at room temperature—experiment versus calculations: a,b) the temporal THz fields of the fundamental driving field and the generated harmonics; experiment in (a) and calculations in (b). c) The dependence of the generated higher-harmonic field on the THz driving field; symbols for the experiment, solid black lines for the calculations, and dashed lines for the power-law fit of the low-signal nonlinearity regime below a driving field strength of 20 kV cm<sup>-1</sup> (Reproduced with permission.<sup>[63]</sup> Copyright 2018, Springer Nature).

the THz nonlinearity in graphene to single-cycle<sup>[59]</sup> and multicycle<sup>[63]</sup> excitation, one can use it to predict the generation of THz harmonics of even higher orders. In **Figure 16**, the calculated spectra of response of graphene to the excitation at the fundamental frequency f = 0.68 THz are shown, predicting the appearance of harmonics up to the 13th order within the dynamic range of 10<sup>6</sup>, characteristic of the TELBE THz facility. The driving field amplitude in this calculation had a maximum of 1 MV cm<sup>-1</sup>, which is a conservative estimate of the field strength required to enter the ballistic nonlinearity regime of coherent Bloch oscillations (see estimate provided in Section 4.1). We note here again, that the HHG output via the collective thermodynamic response of graphene electrons can



**Figure 16.** Thermodynamic model calculations of THz high-harmonic generation up to the 13th order. Calculated spectrum showing high harmonics up to the 13th order within a dynamic range of 10<sup>6</sup> with respect to the transmitted field spectrum at the fundamental frequency (considering a fundamental frequency of 0.68 THz) (Reproduced with permission.<sup>[63]</sup> Copyright 2018, Springer Nature).

be observed at the fields as weak as only 10s of kV cm<sup>-1</sup>, which is 2–3 orders of magnitude weaker than that typically required for initiation of the coherent Bloch oscillation mechanism.

### 5. Conclusions and Future Perspectives

In this Progress Report, we have described the birth and evolution of the research area of THz nonlinear optics of graphene. We have shown that the strongest nonlinear response in graphene is facilitated via an intraband conductivity channel, and that the underlying physical mechanism is a very powerful thermodynamic, collective response of its free carrier population to the driving THz fields. We have summarized the fundamental aspects of a thermodynamic model of the THz nonlinearity of graphene, first described in the work by Mics et al.,<sup>[59]</sup> which rests on basic conservation laws, and relies on perfect conversion of the THz energy absorbed via the intraband conductivity mechanism into electronic heat of the free electron population of graphene. This mechanism explains and quantitatively reproduces the experimentally observed strong THz saturable absorption and nonlinear wave conversion, including extremely efficient THz high harmonics generation in graphene at room temperature and under ambient conditions. As we show, graphene at THz frequencies possesses possibly the highest nonlinear coefficients of all materials we know to date.

We furthermore show that the thermodynamic mechanism underlying the enormous THz nonlinearity in graphene is active at rather moderate driving fields of only 10s of kV cm<sup>-1</sup>. This THz field strength is an order of magnitude lower than the typical channel field of modern high-speed field-effect transistors, routinely operating at 100s of GHz frequencies. This is of high technological relevance since graphene is easily compatible with electronic technologies such as CMOS.<sup>[150,151]</sup> This creates the possibility to utilize CMOS-based sub-THz electronics as efficient and cost-effective pump sources for hybrid devices for



THz frequency synthesis in graphene via the thermodynamic HHG mechanism. We further note that the observed extremely strong nonlinearities, in particular the conversion of as much as about 1% of the pump THz field into the higher THz harmonics, were observed in standard single-layer CVD-grown commercially available graphene sample. Optimization of the light-matter interaction, for example, by creation of stratified structures or metamaterials, will further increase the nonlinear THz output of graphene, leading to even higher conversion efficiencies, and hence to practical graphene-based devices for ultrafast THz signal processing and manipulation. Graphene and hybrid CMOS-graphene-based structures for THz generation via HHG mechanism are considered quite competitive also in comparison with the present generation of electronic diode-based mixing technology.<sup>[152-156]</sup> While the diode-based mixers efficiently operate in the frequency range from a few tens of GHz up to THz,<sup>[156]</sup> graphene-based mixers can potentially offer significantly higher operation bandwidth. Indeed, various modifications of thermodynamic nonlinearity in graphene, varying from the basic HHG effect in unpatterned graphene<sup>[59,63]</sup> to combined thermodynamic-plasmonic nonlinearity in graphene ribbons,<sup>[116]</sup> should allow for the implementation of highly efficient HHGbased frequency synthesis over the entire THz window ranging from 100s of GHz to  $\approx$ 10 THz, and potentially even beyond.

### Acknowledgements

The authors would like to thank Martin Mittendorff for fruitful discussions and for the data of Figure 10b. D.T. acknowledges the project "Nonequilibrium dynamics in solids probed by terahertz fields" funded by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation) (Projektnummer 278162697-SFB 1242). K.J.T. acknowledges funding from the European Union's Horizon 2020 research and innovation programme under Grant Agreement No. 804349 and acknowledges financial support through the MAINZ Visiting Professorship. ICN2 was supported by the Severo Ochoa program from Spanish MINECO (Grant No. SEV-2017-0706). M.G. acknowledges support from the European Cluster of Advanced Laser Light Sources (EUCALL) project which received funding from the European Union's Horizon 2020 research and innovation program under Grant Agreement No. 654220.

# **Conflict of Interest**

The authors declare no conflict of interest.

### **Keywords**

graphene, high-harmonics generation, terahertz frequencies, thermodynamic model, ultrafast nonlinear optics

Received: May 8, 2019 Revised: August 23, 2019 Published online: November 4, 2019

 K. S. Novoselov, A. K. Geim, S. V Morozov, D. Jiang, Y. Zhang, S. V Dubonos, I. V Grigorieva, A. A. Firsov, *Science* 2004, *306*, 666.

- [2] K. S. Novoselov, A. K. Geim, S. V Morozov, D. Jiang, M. I. Katsnelson, I. V Grigorieva, S. V Dubonos, A. A. Firsov, *Nature* 2005, 438, 197.
- [3] A. K. Geim, K. S. Novoselov, Nat. Mater. 2007, 6, 183.
- [4] C. Berger, Z. Song, T. Li, X. Li, A. Y. Ogbazghi, R. Feng, Z. Dai, A. N. Marchenkov, E. H. Conrad, P. N. First, W. A. De Heer, J. Phys. Chem. B 2004, 108, 19912.
- [5] R. R. Nair, P. Blake, A. N. Grigorenko, K. S. Novoselov, T. J. Booth, T. Stauber, N. M. R. Peres, A. K. Geim, *Science* **2008**, *320*, 1308.
- [6] K. F. Mak, M. Y. Sfeir, Y. Wu, C. H. Lui, J. A. Misewich, T. F. Heinz, *Phys. Rev. Lett.* 2008, 101, 196405.
- [7] D. Sun, Z. K. Wu, C. Divin, X. Li, C. Berger, W. A. De Heer, P. N. First, T. B. Norris, *Phys. Rev. Lett.* **2008**, 101, 157402.
- [8] H. Choi, F. Borondics, D. A. Siegel, S. Y. Zhou, M. C. Martin, A. Lanzara, R. A. Kaindl, Appl. Phys. Lett. 2009, 94, 172102.
- [9] S. Winnerl, M. Orlita, P. Plochocka, P. Kossacki, M. Potemski, T. Winzer, E. Malic, A. Knorr, M. Sprinkle, C. Berger, W. A. de Heer, H. Schneider, M. Helm, *Phys. Rev. Lett.* **2011**, *107*, 237401.
- [10] K. M. Dani, J. Lee, R. Sharma, A. D. Mohite, C. M. Galande, P. M. Ajayan, A. M. Dattelbaum, H. Htoon, A. J. Taylor, R. P. Prasankumar, *Phys. Rev. B* **2012**, *86*, 125403.
- [11] C. J. Docherty, M. B. Johnston, J. Infrared, Millimeter, Terahertz Waves **2012**, *33*, 797.
- [12] K. F. Mak, L. Ju, F. Wang, T. F. Heinz, Solid State Commun. 2012, 152, 1341.
- [13] J. M. Dawlaty, S. Shivaraman, J. Strait, P. George, M. Chandrashekhar, Y. Chen, F. Rana, M. G. Spencer, D. Veksler, Y. Chen, *Appl. Phys. Lett.* **2008**, *93*, 131905.
- [14] G. Jnawali, Y. Rao, H. Yan, T. F. Heinz, Nano Lett. 2013, 13, 524.
- [15] A. J. Frenzel, C. H. Lui, W. Fang, N. L. Nair, P. K. Herring, P. Jarillo-Herrero, J. Kong, N. Gedik, *Appl. Phys. Lett.* **2013**, *102*, 113111.
- [16] S. A. Mikhailov, Europhys. Lett. 2007, 79, 27002.
- [17] S. A. Mikhailov, K. Ziegler, J. Phys.: Condens. Matter 2008, 20, 384204.
- [18] A. R. Wright, X. G. Xu, J. C. Cao, C. Zhang, Appl. Phys. Lett. 2009, 95, 072101.
- [19] S. A. Mikhailov, Microelectron. J. 2009, 40, 712.
- [20] Y. S. Ang, S. Sultan, A. Tawfiq, J. Cao, C. Zhang, J. Infrared, Millimeter, Terahertz Waves 2012, 33, 816.
- [21] K. L. Ishikawa, Phys. Rev. B 2010, 82, 201402.
- [22] I. Al-Naib, J. E. Sipe, M. M. Dignam, *Phys. Rev. B* **2014**, *90*, 245423.
- [23] I. Al-Naib, M. Poschmann, M. M. Dignam, Phys. Rev. B 2015, 91, 205407.
- [24] J. L. Cheng, N. Vermeulen, J. E. Sipe, Phys. Rev. B 2015, 91, 235320.
- [25] S. A. Mikhailov, Phys. Rev. B 2017, 95, 085432.
- [26] A. H. Castro Neto, F. Guinea, N. M. R. Peres, K. S. Novoselov, A. K. Geim, *Rev. Mod. Phys.* **2009**, *81*, 109.
- [27] D. S. L. Abergel, V. Apalkov, J. Berashevich, K. Zieglerc, T. Chakraborty, Adv. Phys. 2010, 59, 261.
- [28] S. Das Sarma, S. Adam, E. H. Hwang, E. Rossi, *Rev. Mod. Phys.* 2011, 83, 407.
- [29] Q. Bao, H. Zhang, Y. Wang, Z. Ni, Y. Yan, Z. X. Shen, K. P. Loh, D. Y. Tang, Adv. Funct. Mater. 2009, 19, 3077.
- [30] Z. Sun, T. Hasan, F. Torrisi, D. Popa, G. Privitera, F. Wang, F. Bonaccorso, D. M. Basko, A. C. Ferrari, ACS Nano 2010, 4, 803.
- [31] N. Vermeulen, D. Castelló-lurbe, M. Khoder, I. Pasternak, A. Krajewska, T. Ciuk, W. Strupinski, J. Cheng, H. Thienpont, J. Van Erps, *Nat. Commun.* **2018**, *9*, 2675.
- [32] D. Popa, Z. Sun, F. Torrisi, T. Hasan, F. Wang, A. C. Ferrari, Appl. Phys. Lett. 2010, 97, 203106.
- [33] G.-K. Lim, Z.-L. Chen, J. Clark, R. G. S. Goh, W.-H. Ng, H.-W. Tan, R. H. Friend, P. K. H. Ho, L.-L. Chua, *Nat. Photonics* 2011, 5, 554.

#### **ADVANCED** SCIENCE NEWS

www.advancedsciencenews.com

- [34] D. Popa, Z. Sun, T. Hasan, F. Torrisi, F. Wang, A. C. Ferrari, Appl. Phys. Lett. 2011, 98, 073106.
- [35] H. Zhang, S. Virally, Q. Bao, L. K. Ping, S. Massar, N. Godbout, P. Kockaert, Opt. Lett. 2012, 37, 1856.
- [36] W. Chen, G. Wang, S. Qin, C. Wang, J. Fang, J. Qi, X. Zhang, L. Wang, H. Jia, S. Chang, AIP Adv. 2013, 3, 042123.
- [37] G. Demetriou, H. T. Bookey, F. Biancalana, E. Abraham, W. Ji, A. K. Kar, Opt. Express 2016, 24, 13033.
- [38] N. Vermeulen, D. Castelló-Lurbe, J. Cheng, I. Pasternak, A. Krajewska, T. Ciuk, W. Strupinski, H. Thienpont, J. Van Erps, *Phys. Rev. Appl.* **2016**, *6*, 044006.
- [39] K. Alexander, N. A. Savostianova, S. A. Mikhailov, D. Van Thourhout, B. Kuyken, ACS Photonics 2018, 5, 4944.
- [40] A. Y. Bykov, T. V. Murzina, M. G. Rybin, E. D. Obraztsova, *Phys. Rev. B* 2012, *85*, 121413.
- [41] Y. Q. An, F. Nelson, J. U. Lee, A. C. Diebold, Nano Lett. 2013, 13, 2104.
- [42] S. Hong, J. I. Dadap, N. Petrone, P. Yeh, J. Hone, R. M. Osgood, *Phys. Rev. X* 2013, 3, 021014.
- [43] N. Kumar, J. Kumar, C. Gerstenkorn, R. Wang, H. Chiu, A. L. Smirl, H. Zhao, Phys. Rev. B 2013, 87, 121406.
- [44] Y. Q. An, J. E. Rowe, D. B. Dougherty, J. U. Lee, A. C. Diebold, *Phys. Rev. B* 2014, *89*, 115310.
- [45] K. H. Lin, S. W. Weng, P. W. Lyu, T. R. Tsai, W. Bin Su, Appl. Phys. Lett. 2014, 105, 151605.
- [46] G. Soavi, G. Wang, H. Rostami, D. G. Purdie, D. De Fazio, T. Ma, B. Luo, J. Wang, A. K. Ott, D. Yoon, S. A. Bourelle, J. E. Muench, I. Goykhman, S. Dal Conte, M. Celebrano, A. Tomadin, M. Polini, G. Cerullo, A. C. Ferrari, *Nat. Nanotechnol.* **2018**, *13*, 583.
- [47] N. Yoshikawa, T. Tamaya, K. Tanaka, Science 2017, 356, 736.
- [48] E. Hendry, P. J. Hale, J. J. Moger, A. K. Savchenko, S. A. Mikhailov, *Phys. Rev. Lett.* **2009**, *105*, 097401.
- [49] T. Gu, N. Petrone, J. F. McMillan, A. van der Zande, M. Yu, G. Q. Lo, D. L. Kwong, J. Hone, C. W. Wong, *Nat. Photonics* 2012, 6, 554.
- [50] K. Alexander, N. A. Savostianova, S. A. Mikhailov, B. Kuyken, D. Van Thourhout, ACS Photonics 2017, 4, 3039.
- [51] K. J. A. Ooi, D. T. H. Tan, Proc. R. Soc. A 2017, 473, 20170433.
- [52] A. Autere, H. Jussila, Y. Dai, Y. Wang, H. Lipsanen, Z. Sun, Adv. Mater. 2018, 30, 1705963.
- [53] J. W. You, S. R. Bongu, Q. Bao, N. C. Panoiu, *Nanophotonics* 2018, 8, 63.
- [54] H. Y. Hwang, N. C. Brandt, H. Farhat, A. L. Hsu, J. Kong, K. A. Nelson, arXiv:1101.4985, 2011.
- [55] H. Y. Hwang, N. C. Brandt, H. Farhat, A. L. Hsu, J. Kong, K. A. Nelson, J. Phys. Chem. B 2013, 117, 15819.
- [56] M. J. Paul, Y. C. Chang, Z. J. Thompson, A. Stickel, J. Wardini, H. Choi, E. D. Minot, T. B. Norris, Y.-S. Lee, *New J. Phys.* **2013**, *15*, 085019.
- [57] H. A. Hafez, I. Al-Naib, K. Oguri, Y. Sekine, M. M. Dignam, A. Ibrahim, D. G. Cooke, S. Tanaka, F. Komori, H. Hibino, T. Ozaki, AIP Adv. 2014, 4, 117118.
- [58] M. J. Paul, B. Lee, J. L. Wardini, Z. J. Thompson, A. D. Stickel, A. Mousavian, H. Choi, E. D. Minot, Y.-S. Lee, *Appl. Phys. Lett.* 2014, 105, 221107.
- [59] Z. Mics, K.-J. Tielrooij, K. Parvez, S. A. Jensen, I. Ivanov, X. Feng, K. Müllen, M. Bonn, D. Turchinovich, *Nat. Commun.* 2015, *6*, 7655.
- [60] T. Elsaesser, K. Reimann, M. Werner, Concepts and Applications of Nonlinear Terahertz Spectroscopy, Morgan & Claypool Publishers, San Rafael, CA 2019.
- [61] H. A. Hafez, X. Chai, A. Ibrahim, S. Mondal, D. Férachou, X. Ropagnol, T. Ozaki, J. Opt. 2016, 18, 093004.
- [62] P. Bowlan, E. Martinez-Moreno, K. Reimann, T. Elsaesser, M. Woerner, Phys. Rev. B 2014, 89, 041408.



- [63] H. A. Hafez, S. Kovalev, J.-C. Deinert, Z. Mics, B. Green, N. Awari, M. Chen, S. Germanskiy, U. Lehnert, J. Teichert, Z. Wang, K.-J. Tielrooij, Z. Liu, Z. Chen, A. Narita, K. Müllen, M. Bonn, M. Gensch, D. Turchinovich, *Nature* **2018**, *561*, 507.
- [64] F. Schwierz, Nat. Nanotechnol. 2010, 5, 487.
- [65] R. Cheng, J. Bai, L. Liao, H. Zhou, Y. Chen, L. Liu, Y.-C. Lin, S. Jiang, Y. Huang, X. Duan, *Proc. Natl. Acad. Sci. USA* **2012**, 109, 11588.
- [66] F. Schwierz, Proc. IEEE 2013, 101, 1567.
- [67] F. Schwierz, J. Pezoldt, R. Granzner, Nanoscale 2015, 7, 8261.
- [68] T. Mueller, F. Xia, P. Avouris, Nat. Photonics 2010, 4, 297.
- [69] M. Mittendorff, S. Winnerl, J. Kamann, J. Eroms, D. Weiss, H. Schneider, M. Helm, Appl. Phys. Lett. 2013, 103, 021113.
- [70] V. Mamidala, Y. Yang, H. H. Liu, K. K. Chow, R. I. Woodward, Opt. Express 2014, 22, 4539.
- [71] G. Sobon, J. Sotor, I. Pasternak, A. Krajewska, W. Strupinski, K. M. Abramski, Opt. Mater. Express 2015, 5, 2884.
- [72] J. Basu, J. K. Basu, T. K. Bhattacharyya, Int. J. Smart Nano Mater. 2010, 1, 201.
- [73] P. R. Wallace, Phys. Rev. 1947, 71, 622.
- [74] G. Wang, A. Chernikov, M. M. Glazov, T. F. Heinz, X. Marie, T. Amand, B. Urbaszek, *Rev. Mod. Phys.* **2018**, *90*, 021001.
- [75] A. S. Mayorov, R. V. Gorbachev, S. V. Morozov, L. Britnell, R. Jalil, L. A. Ponomarenko, P. Blake, K. S. Novoselov, K. Watanabe, T. Taniguchi, A. K. Geim, *Nano Lett.* **2011**, *11*, 2396.
- [76] K. F. Mak, J. Shan, T. F. Heinz, Phys. Rev. Lett. 2011, 106, 046401.
- [77] P. Blake, E. W. Hill, Appl. Phys. Lett. 2007, 91, 063124.
- [78] K. J. Tielrooij, L. Piatkowski, M. Massicotte, A. Woessner, Q. Ma, Y. Lee, K. S. Myhro, C. N. Lau, P. Jarillo-Herrero, N. F. Van Hulst, F. H. L. Koppens, *Nat. Nanotechnol.* **2015**, *10*, 437.
- [79] F. Wang, Y. Zhang, C. Tian, C. Girit, A. Zettl, M. Crommie, Y. R. Shen, *Science* **2008**, *320*, 206.
- [80] Z. Q. Li, E. A. Henriksen, Z. Jiang, Z. Hao, M. C. Martin, P. Kim, H. L. Stormer, D. N. Basov, *Nat. Phys.* **2008**, *4*, 532.
- [81] X. Gan, R. J. Shiue, Y. Gao, K. F. Mak, X. Yao, L. Li, A. Szep, D. Walker, J. Hone, T. F. Heinz, D. Englund, *Nano Lett.* **2013**, *13*, 691.
- [82] I. Ivanov, M. Bonn, Z. Mics, D. Turchinovich, EPL 2015, 111, 67001.
- [83] Y. Zhang, V. W. Brar, C. Girit, A. Zettl, M. F. Crommie, Nat. Phys. 2009, 5, 722.
- [84] S. A. A. Jensen, Z. Mics, I. Ivanov, H. S. S. Varol, D. Turchinovich, F. H. L. H. L. Koppens, M. Bonn, K. J. J. Tielrooij, *Nano Lett.* 2014, 14, 5839.
- [85] H. I. Wang, M.-L. Braatz, N. Richter, K.-J. Tielrooij, Z. Mics, H. Lu, N.-E. Weber, K. Müllen, D. Turchinovich, M. Kläui, M. Bonn, J. Phys. Chem. C 2017, 121, 4083.
- [86] H. A. Hafez, X. Chai, Y. Sekine, M. Takamura, K. Oguri, I. Al-Naib, M. M. Dignam, H. Hibino, T. Ozaki, *Phys. Rev. B* 2017, 95, 165428.
- [87] F. Schedin, A. K. Geim, S. V. Morozov, E. W. Hill, P. Blake, M. I. Katsnelson, K. S. Novoselov, *Nat. Mater.* 2007, *6*, 652.
- [88] C. Coletti, C. Riedl, D. S. Lee, B. Krauss, L. Patthey, K. Von Klitzing, J. H. Smet, U. Starke, *Phys. Rev. B* 2010, *81*, 235401.
- [89] C. J. Docherty, C.-T. Lin, H. J. Joyce, R. J. Nicholas, L. M. Herz, L.-J. Li, M. B. Johnston, *Nat. Commun.* 2012, *3*, 1228.
- [90] A. Pirkle, J. Chan, A. Venugopal, D. Hinojos, C. W. Magnuson, S. McDonnell, L. Colombo, E. M. Vogel, R. S. Ruoff, R. M. Wallace, *Appl. Phys. Lett.* **2011**, *99*, 122108.
- [91] S. Ryu, L. Liu, S. Berciaud, Y.-J. Yu, H. Liu, P. Kim, G. W. Flynn, L. E. Brus, *Nano Lett.* **2010**, *10*, 4944.
- [92] H. Hibino, H. Kageshima, M. Nagase, J. Phys. D: Appl. Phys. 2010, 43, 374005.

#### **ADVANCED** SCIENCE NEWS

www.advancedsciencenews.com



- [93] J. C. König-Otto, M. Mittendorff, T. Winzer, F. Kadi, E. Malic, A. Knorr, C. Berger, W. A. De Heer, A. Pashkin, H. Schneider, M. Helm, S. Winnerl, *Phys. Rev. Lett.* **2016**, *117*, 087401.
- [94] R. Ulbricht, E. Hendry, J. Shan, T. Heinz, M. Bonn, Rev. Mod. Phys. 2011, 83, 543.
- [95] P. U. Jepsen, D. G. Cooke, M. Koch, Laser Photonics Rev. 2011, 5, 124.
- [96] R. E. Glover, M. Tinkham, Phys. Rev. 1957, 108, 243.
- [97] M. Baudisch, A. Marini, J. D. Cox, T. Zhu, F. Silva, S. Teichmann, M. Massicotte, F. Koppens, L. S. Levitov, F. J. García de Abajo, J. Biegert, *Nat. Commun.* **2018**, *9*, 1018.
- [98] J. C. W. Song, K. J. Tielrooij, F. H. L. Koppens, L. S. Levitov, Phys. Rev. B 2013, 87, 155429.
- [99] D. Brida, A. Tomadin, C. Manzoni, Y. J. Kim, A. Lombardo, S. Milana, R. R. Nair, K. S. Novoselov, A. C. Ferrari, G. Cerullo, M. Polini, *Nat. Commun.* **2013**, *4*, 1987.
- [100] I. Gierz, J. C. Petersen, M. Mitrano, C. Cacho, I. C. E. Turcu,
   E. Springate, A. Stöhr, A. Köhler, U. Starke, A. Cavalleri, *Nat. Mater.* 2013, 12, 1119.
- [101] J. C. W. Song, M. Y. Reizer, L. S. Levitov, Phys. Rev. Lett. 2012, 109, 106602.
- [102] M. T. Mihnev, F. Kadi, C. J. Divin, T. Winzer, S. Lee, C. H. Liu, Z. Zhong, C. Berger, W. A. De Heer, E. Malic, A. Knorr, T. B. Norris, *Nat. Commun.* 2016, 7, 11617.
- [103] T. Low, V. Perebeinos, R. Kim, M. Freitag, P. Avouris, *Phys. Rev. B* 2012, *86*, 045413.
- [104] K.-J. Tielrooij, N. C. H. Hesp, A. Principi, M. B. Lundeberg, E. A. A. Pogna, L. Banszerus, Z. Mics, M. Massicotte, P. Schmidt, D. Davydovskaya, D. G. Purdie, I. Goykhman, G. Soavi, A. Lombardo, K. Watanabe, T. Taniguchi, M. Bonn, D. Turchinovich, C. Stampfer, A. C. Ferrari, G. Cerullo, M. Polini, F. H. L. Koppens, *Nat. Nanotechnol.* **2018**, *13*, 41.
- [105] R. Bistritzer, A. H. MacDonald, Phys. Rev. Lett. 2009, 102, 206410.
- [106] O. Schubert, M. Hohenleutner, F. Langer, B. Urbanek, C. Lange, U. Huttner, D. Golde, T. Meier, M. Kira, S. W. Koch, R. Huber, *Nat. Photonics* **2014**, *8*, 119.
- [107] T. Tamaya, A. Ishikawa, T. Ogawa, K. Tanaka, Phys. Rev. Lett. 2016, 116, 016601.
- [108] T. Higuchi, C. Heide, K. Ullmann, H. B. Weber, P. Hommelhoff, *Nature* 2017, 550, 224.
- [109] A. W. Ghosh, M. C. Wanke, S. J. Allen, J. W. Wilkins, M. C. Wanke, S. J. Allen, J. W. Wilkins, *Appl. Phys. Lett.* **1999**, *74*, 2164.
- [110] S. Winnerl, E. Schomburg, S. Brandl, O. Kus, K. F. Renk, M. C. Wanke, S. J. Allen, A. A. Ignatov, V. Ustinov, A. Zhukov, P. S. Kop'ev, *Appl. Phys. Lett.* **2000**, *77*, 1259.
- [111] H. Razavipour, W. Yang, A. Guermoune, M. Hilke, D. G. Cooke, I. Al-Naib, M. M. Dignam, F. Blanchard, H. A. Hafez, X. Chai, D. Ferachou, T. Ozaki, P. L. Lévesque, R. Martel, *Phys. Rev. B* 2015, 92, 245421.
- [112] E. Dremetsika, P. Kockaert, Phys. Rev. B 2017, 96, 235422.
- [113] D. Kundys, B. Van Duppen, O. P. Marshall, F. Rodriguez, I. Torre, A. Tomadin, M. Polini, A. N. Grigorenko, *Nano Lett.* 2018, 18, 282.
- [114] J. C. König-Otto, Y. Wang, A. Belyanin, C. Berger, W. A. De Heer, M. Orlita, A. Pashkin, H. Schneider, M. Helm, S. Winnerl, *Nano Lett.* 2017, *17*, 2184.
- [115] M. M. Jadidi, J. C. König-Otto, S. Winnerl, A. B. Sushkov, H. D. Drew, T. E. Murphy, M. Mittendorff, *Nano Lett.* **2016**, *16*, 2734.
- [116] M. M. Jadidi, A. B. Sushkov, D. K. Gaskill, T. E. Murphy, S. Winnerl, J. C. König-Otto, R. L. Myers-Ward, M. Mittendorff, K. M. Daniels, H. D. Drew, ACS Photonics 2019, 6, 302.
- [117] C. Sirtori, F. Capasso, D. L. Sivco, A. Y. Cho, A. T. B. I, M. Hill, *Phys. Rev. Lett.* **1992**, *68*, 1010.
- [118] J. Kono, M. Y. Su, T. Inoshita, T. Noda, M. S. Sherwin, S. J. Allen, H. Sakaki, *Phys. Rev. Lett.* **1997**, *79*, 1758.

- [119] H. A. Hafez, I. Al-Naib, M. M. Dignam, Y. Sekine, K. Oguri,
   F. Blanchard, D. G. Cooke, S. Tanaka, F. Komori, H. Hibino,
   T. Ozaki, *Phys. Rev. B* 2015, *91*, 035422.
- [120] H. A. Hafez, P. L. Lévesque, I. Al-Naib, M. M. Dignam, X. Chai, S. Choubak, P. Desjardins, R. Martel, T. Ozaki, *Appl. Phys. Lett.* 2015, 107, 251903.
- [121] S. Tani, F. Blanchard, K. Tanaka, Phys. Rev. Lett. 2012, 109, 166603.
- [122] T. Kampfrath, K. Tanaka, K. A. Nelson, Nat. Photonics 2013, 7, 680.
- S. A. Zvyagin, M. Ozerov, E. Čižmár, D. Kamenskyi, S. Zherlitsyn, T. Herrmannsdörfer, J. Wosnitza, R. Wünsch, W. Seidel, *Rev. Sci. Instrum.* 2009, *80*, 073102.
- [124] B. Green, S. Kovalev, V. Asgekar, G. Geloni, U. Lehnert, T. Golz, M. Kuntzsch, C. Bauer, J. Hauser, J. Voigtlaender, B. Wustmann, I. Koesterke, M. Schwarz, M. Freitag, A. Arnold, J. Teichert, M. Justus, W. Seidel, C. Ilgner, N. Awari, D. Nicoletti, S. Kaiser, Y. Laplace, S. Rajasekaran, L. Zhang, S. Winnerl, H. Schneider, G. Schay, I. Lorincz, A. A. Rauscher, I. Radu, S. Mährlein, T. H. Kim, J. S. Lee, T. Kampfrath, S. Wall, J. Heberle, A. Malnasi-Csizmadia, A. Steiger, A. S. Müller, M. Helm, U. Schramm, T. Cowan, P. Michel, A. Cavalleri, A. S. Fisher, N. Stojanovic, M. Gensch, *Sci. Rep.* 2016, *6*, 22256.
- [125] M. Mittendorff, F. Wendler, E. Malic, A. Knorr, M. Orlita, M. Potemski, C. Berger, W. A. de Heer, H. Schneider, M. Helm, S. Winnerl, *Nat. Phys.* **2015**, *11*, 75.
- [126] S. Winnerl, M. Mittendorff, J. C. König-Otto, H. Schneider, M. Helm, T. Winzer, A. Knorr, E. Malic, Ann. Phys. 2017, 529, 1700022.
- [127] M. Mittendorff, T. Winzer, E. Malić, A. Knorr, C. Berger, W. A. de Heer, H. Schneider, M. Helm, S. Winnerl, *Nano Lett.* 2014, 14, 1504.
- [128] A. A. Melnikov, A. A. Sokolik, A. V Frolov, S. V Chekalin, E. A. Ryabov, *Appl. Phys. Lett.* **2019**, *114*, 191107.
- [129] K. J. Tielrooij, J. C. W. Song, S. A. Jensen, A. Centeno, A. Pesquera,
   A. Z. Elorza, M. Bonn, L. S. Levitov, F. H. L. Koppens, *Nat. Phys.* **2013**, *9*, 248.
- [130] C. H. Lui, K. F. Mak, J. Shan, T. F. Heinz, Phys. Rev. Lett. 2010, 105, 127404.
- [131] S.-F. Shi, T.-T. Tang, B. Zeng, L. Ju, Q. Zhou, A. Zettl, F. Wang, Nano Lett. 2014, 14, 1578.
- [132] A. J. Frenzel, C. H. Lui, Y. C. Shin, J. Kong, N. Gedik, Phys. Rev. Lett. 2014, 113, 056602.
- [133] A. Tomadin, S. M. Hornett, H. I. Wang, E. M. Alexeev, A. Candini, C. Coletti, D. Turchinovich, M. Kläui, M. Bonn, F. H. L. Koppens, E. Hendry, M. Polini, K.-J. Tielrooij, *Sci. Adv.* 2018, *4*, eaar5313.
- [134] T. Winzer, A. Knorr, E. Malic, Nano Lett. 2010, 10, 4839.
- [135] T. Plötzing, T. Winzer, E. Malic, D. Neumaier, A. Knorr, H. Kurz, Nano Lett. 2014, 14, 5371.
- [136] J. D. Cox, F. J. García De Abajo, ACS Photonics 2015, 2, 306.
- [137] J. D. Cox, A. Marini, F. J. G. de Abajo, Nat. Commun. 2017, 8, 14380.
- [138] J. D. Cox, F. J. García de Abajo, Optica 2018, 5, 429.
- [139] A. Tomadin, D. Brida, G. Cerullo, A. C. Ferrari, M. Polini, *Phys. Rev. B* 2013, *88*, 035430.
- [140] G. Moos, C. Gahl, R. Fasel, M. Wolf, T. Hertel, Phys. Rev. Lett. 2001, 87, 267402.
- [141] W.-K. Tse, E. H. Hwang, S. Das Sarma, Appl. Phys. Lett. 2008, 93, 023128.
- [142] E. H. Hwang, B. Y.-K. Hu, S. Das Sarma, Phys. Rev. B 2007, 76, 115434.
- [143] M. Breusing, S. Kuehn, T. Winzer, E. Malić, F. Milde, N. Severin, J. P. Rabe, C. Ropers, A. Knorr, T. Elsaesser, *Phys. Rev. B* 2011, *83*, 153410.
- [144] Z. Mics, A. D'Angio, S. A. Jensen, M. Bonn, D. Turchinovich, Appl. Phys. Lett. 2013, 102, 231120.

#### **ADVANCED** SCIENCE NEWS

www.advancedsciencenews.com



- [145] X. Liu, J. Lægsgaard, D. Turchinovich, Opt. Lett. 2010, 35, 913.
- [146] X. Liu, J. Laegsgaard, R. Iegorov, A. S. Svane, F. Ö. Ilday, H. Tu, S. A. Boppart, D. Turchinovich, *Photonics Res.* 2017, 5, 750.
- [147] I. Gierz, F. Calegari, S. Aeschlimann, M. Chávez Cervantes, C. Cacho, R. T. Chapman, E. Springate, S. Link, U. Starke, C. R. Ast, A. Cavalleri, *Phys. Rev. Lett.* **2015**, *115*, 086803.
- [148] J. C. Johannsen, S. Ulstrup, F. Cilento, A. Crepaldi, M. Zacchigna, C. Cacho, I. C. E. Turcu, E. Springate, F. Fromm, C. Raidel, T. Seyller, F. Parmigiani, M. Grioni, P. Hofmann, *Phys. Rev. Lett.* **2013**, *111*, 027403.
- [149] M. Breusing, C. Ropers, T. Elsaesser, Phys. Rev. Lett. 2009, 102, 086809.
- [150] S. Goossens, G. Navickaite, C. Monasterio, S. Gupta, J. J. Piqueras, R. Pérez, G. Burwell, I. Nikitskiy, T. Lasanta, T. Galán,

E. Puma, A. Centeno, A. Pesquera, A. Zurutuza, G. Konstantatos, F. Koppens, *Nat. Photonics* **2017**, *11*, 366.

- [151] D. Neumaier, S. Pindl, M. C. Lemme, Nat. Mater. 2019, 18, 525.
- [152] V. G. Bozhkov, Radiophys. Quantum Electron. 2003, 46, 631.
- [153] J. V. Siles, J. Grajal, *IEEE Trans. Microwave Theory Tech.* **2010**, *58*, 1933.
- [154] A. Maestrini, J. S. Ward, J. J. Gill, C. Lee, B. Thomas, R. H. Lin, G. Chattopadhyay, I. Mehdi, *IEEE Trans. Microwave Theory Tech.* 2010, 58, 1925.
- [155] D. G. Paveliev, Y. I. Koshurinov, S. Ivanov, A. N. Panin, V. L. Vax, V. I. Gavrilenko, A. V. Antonov, V. M. Ustinov, A. E. Zhukov, *Semiconductors* **2012**, *46*, 121.
- [156] I. Mehdi, J. V. Siles, C. Lee, E. Schlecht, Proc. IEEE 2017, 105, 990.